# COMPETENCES FOR DEMOCRATIC CULTURE

Living together as equals in culturally diverse democratic societies





# COMPETENCES FOR DEMOCRATIC CULTURE

Living together as equals in culturally diverse democratic societies

The opinions expressed in this work are the responsibility of the authors and do not necessarily reflect the official policy of the Council of Europe..

All rights reserved. No part of this publication may be translated, reproduced or transmitted, in any form or by any means, electronic (CD-Rom, Internet, etc.) or mechanical, including photocopying, recording or any information storage or retrieval system,

(F-67075 Strasbourg Cedex or publishing@coe.int).

without prior permission in writing from the Directorate of Communication

Cover photo: © Shutterstock

Cover design: Documents and

Publications Production Department (SPDP), | Council of Europe | Layout: Documents and Publications Production | Department (SPDP), Council of Europe | Council of Eur

Council of Europe Publishing
F-67075 Strasbourg Cedex
http://book.coe.int

ISBN 978-92-871-8237-1 © Council of Europe, March 2016 Printed at the Council of Europe

## About this document

This document is a product of a Council of Europe project which is taking place in four phases during 2014-17. The first phase has been devoted to the development of a conceptual model of the competences which citizens require to participate effectively in a culture of democracy. This document describes the model and the methods that were used to develop it. The document is aimed at readers who wish to understand the underlying assumptions and technical details of the model.

Phase two of the project will be devoted to the development of descriptors (i.e. statements or descriptions of what a person is able to do if they have mastered the various competences that are specified by the model), phase three to ascertaining whether it is possible to assign the descriptors to levels of proficiency, and phase four to the production of supporting documentation. This documentation will be addressed to educational practitioners and policy makers, and will provide a less technical description of the current competence model. It will also explain how the model and the descriptors can be used to assist curriculum design, pedagogical design and the development of new forms of assessment (for use in either self-assessment or assessment by others).

All of the materials that are produced by the project will eventually be incorporated into a Council of Europe reference framework of competences for democratic culture.

Further information about the project is available from the project website: www.coe.int/competences

## **Contents**

ABOUT THIS DOCUMENT	3
PREFACE	7
EXECUTIVE SUMMARY	9
1. INTRODUCTION	15
2. A BACKGROUND ASSUMPTION: COMPETENCES ARE NECESSARY BUT NOT SUFFICIENT	17
3. A FURTHER BACKGROUND ASSUMPTION: ALL CULTURES ARE INTERNALLY HETEROGENEOUS, CONTESTED, DYNAMIC AND CONSTANTLY EVOLVING	19
4. THE CONCEPT OF COMPETENCE EMPLOYED IN THE CURRENT MODEL	23
5. THE WORKING METHOD USED TO IDENTIFY THE COMPETENCES TO BE INCLUDED IN THE MODEL	27
5.1. Audit of existing competence schemes	27
5.2. Decomposition of existing competence schemes	28
5.3. The identification of competences for inclusion in the model	29
6. THE MODEL	35
6.1. Values	36
6.2. Attitudes	39
6.3. Skills	44
6.4. Knowledge and critical understanding	51
7. CONCLUSION	57
APPENDIX A THE SOURCES OF THE COMPETENCE SCHEMES THAT WERE AUDITED BY THE PROJECT	59
APPENDIX B	
THE 55 POSSIBLE COMPETENCES THAT WERE IDENTIFIED ACROSS THE 101 COMPETENCE SCHEMES	69
APPENDIX C SUGGESTIONS FOR FURTHER READING BEYOND THE REFERENCES LISTED IN APPENDIX A	71
APPENDIX D AUTHORS AND CONTRIBUTORS	75

## **Preface**

he Council of Europe has long been active in the fields of education for democratic citizenship and intercultural dialogue. Our European Cultural Convention first entered into force in 1954. While the memory of war was still stalking the continent and new divisions were taking hold, Europe's nations states agreed, through this Treaty, to encourage the study of languages, history and civilisation for the sake of unity: to help "safeguard and realise the ideals and principles which are [our] common heritage".

Ever since, the Council of Europe has been able to play a leading role in re-establishing trust and understanding through education in many of the conflict situations which have arisen in Europe. Through education we have also been able to assist many newer member states in their transitions to democracy. Democracy is, of course, built on institutions and laws, but it lives through the actions and behaviour of its citizens. Democratic culture must therefore be taught and fostered too.

Today Europe's nations face new challenges which require our Organisation to step up our support in the classroom. Increased migration, growing diversity, the boom in information technology and globalisation are having a profound effect on people's identities. More than ever, within our communities we find people living side-by-side who hold different beliefs, backgrounds and outlooks. This enriching of European societies is to be celebrated, but it also requires us to think carefully about how we nurture a set of common values around which to organise. How do we resolve clashes between competing worldviews? What are the attitudes and behaviours we can and cannot accept?

Such dilemmas are not easy. The danger of avoiding these questions, however, has been brought sharply into focus by the recent surge in foreign terrorist fighters: young, radicalised Europeans who have been brainwashed into turning their back on democratic life and waging war on their fellow citizens. Such extremism can only take root when young minds have not been taught to understand diversity, rather than to fear it, and when young people struggle to think critically, for themselves.

The Council of Europe is therefore equipping the continent's educators with a ground-breaking set of competences to help teach pupils how to live together, as democratic citizens in diverse societies. As our nations continue to grow more mixed, such knowledge and skills will become ever more important. The aim is not to teach students what to think, but rather how to think, in order to navigate a world where not everyone holds their views, but we each have a duty to uphold the democratic principles which allow all cultures to co-exist.

The competences have been developed in such a way as to allow member states to adapt them to suit their own needs and the distinct cultural contours of their own societies. Through this framework, teachers will be able to instill in their pupils the values of tolerance and respect, as they grow to understand their rights and responsibilities in relation to others. We will continue to educate for democracy by offering our member states advice on the implementation of this competence framework and help in training those who will make it work in practice. I hope that all member states will give this ambitious and timely initiative their full support.

Thorbjørn Jagland Secretary General of the Council of Europe

# **Executive summary**

his document describes a conceptual model of the competences which need to be acquired by learners if they are to participate effectively in a culture of democracy and live peacefully together with others in culturally diverse democratic societies. It is intended that the model will be used to inform educational decision making and planning, helping educational systems to be harnessed for the preparation of learners for life as competent democratic citizens.

The document is divided into seven chapters.

In Chapter 1, the educational purpose of the competence model is outlined. This section also explains why the phrase "culture of democracy" is used in the present context rather than "democracy": this is to emphasise the fact that, while democracy cannot exist without democratic institutions and laws, such institutions and laws cannot work in practice unless they are grounded in a culture of democracy, that is, in democratic values, attitudes and practices. Chapter 1 also explains the interdependence between a culture of democracy and intercultural dialogue in culturally diverse societies: in such societies, intercultural dialogue is vital to ensure the inclusion of all citizens in democratic discussion, debate and deliberation.

Chapters 2 and 3 describe some of the background assumptions underlying the model. Chapter 2 describes the assumption that, while it is necessary for citizens to acquire a range of competences in order to participate effectively in a culture of democracy, these competences are not sufficient for such participation to occur because democratic participation also requires appropriate institutional structures. In other words, both competences and democratic institutions are essential to sustain a culture of democracy. In addition, the democratic participation of all citizens within society requires measures to tackle social inequalities and structural disadvantages. In the absence of such measures, the members of disadvantaged groups will be marginalised in democratic processes, whatever their levels of democratic competence might be.

Chapter 3 describes the concept of "culture" that is assumed by the competence model. All cultures are internally heterogeneous, contested, dynamic and constantly evolving, and all people inhabit multiple cultures that interact in complex ways. The concept of "intercultural" is also examined in this section. It is proposed that intercultural situations arise when an individual perceives another person or group as being culturally different from themselves. Intercultural dialogue is therefore defined as dialogue that takes place between individuals or groups who perceive themselves as having different cultural affiliations from each other. It is noted that, although intercultural dialogue is extremely important for fostering tolerance and enhancing social cohesion in culturally diverse societies, such dialogue can be extremely demanding and difficult in some circumstances.

Chapter 4 then unpacks the concept of "competence" that is employed by the model. Democratic and intercultural competence is defined as the ability to mobilise and deploy relevant values, attitudes, skills, knowledge and/or understanding in order to respond appropriately and effectively to the demands, challenges and opportunities that are presented by democratic and intercultural situations. Competence is treated as a dynamic process in which a competent individual mobilises and deploys clusters of psychological resources in an active and adaptive manner in order to respond to new circumstances as these arise.

Chapter 4 also describes how, in addition to this global and holistic use of the term "competence" (in the singular), the term "competences" (in the plural) is used in the current document to refer to the specific individual resources (i.e. the specific values, attitudes, skills, knowledge and understanding) that are mobilised and deployed in the production of competent behaviour. In other words, in the present account, competence consists of the selection, activation and organisation of competences and the application of these competences in a co-ordinated, adaptive and dynamic manner to concrete situations.

Chapter 5 describes the working method through which specific competences were identified for inclusion in the current model. A notable feature of the model is that it was not designed from scratch. Instead, it was grounded in a systematic analysis of existing conceptual schemes of democratic competence and intercultural competence. An audit was conducted through which 101 such schemes were identified. These 101 schemes were decomposed to identify all the individual competences which they contained, and these competences were then grouped into cognate sets. This led to the identification of 55 possible competences for inclusion in the model. In order to assist in reducing this list of competences to a more manageable and practical length, a set of principled criteria and pragmatic considerations was used to identify the key competences which needed to be included within the model. The application of these criteria and considerations led to the identification of 20 competences for inclusion in the model: 3 sets of values, 6 attitudes, 8 skills and 3 bodies of knowledge and critical understanding. These competences were used to construct the model. A draft document describing the model was then produced and circulated in an international consultation exercise involving academic experts, educational practitioners and policy makers. The responses received in the consultation strongly endorsed the model but also provided a range of useful feedback. The feedback was used to fine-tune the details of the model and to guide the writing of the current document.

#### The 20 competences included in the model

#### **Values**

- Valuing human dignity and human rights
- Valuing cultural diversity
- Valuing democracy, justice, fairness, equality and the rule of law

#### **Attitudes**

- Openness to cultural otherness and to other beliefs, world views and practices
- Respect
- Civic-mindedness
- Responsibility
- Self-efficacy
- Tolerance of ambiguity

#### **Competence**

- Autonomous learning skills
- Analytical and critical thinking skills
- Skills of listening and observing
- Empathy
- Flexibility and adaptability
- Linguistic, communicative and plurilingual skills
- Co-operation skills
- Conflict-resolution skills

#### **Skills**

- Knowledge and critical understanding of the self
- Knowledge and critical understanding of language and communication
- Knowledge and critical understanding of the world: politics, law, human rights, culture, cultures, religions, history, media, economies, environment, sustainability

Knowledge and critical understanding

Chapter 6 describes the resulting model in full, by listing and describing all of the specific values, attitudes, skills, knowledge and critical understanding which enable an individual to participate effectively and appropriately in a culture of democracy. The model is summarised in the diagram above, while a full list of the 20 competences, together with a summary description of each competence, is provided in the box below.

Chapter 7 concludes the document by noting two hopes for the current model: that it will prove useful for educational decision making and planning, and that it will assist in the empowerment of young people as autonomous social agents capable of choosing and pursuing their own goals in life within the framework that is provided by democratic institutions and respect for human rights.

Appendix A provides a list of the sources of the 101 competence schemes that were audited by the project. Appendix B provides a list of the 55 possible competences that were identified across the 101 schemes. Appendix C provides some suggestions for further reading beyond the references that are listed in Appendix A.

# A summary list of the competences which enable an individual to participate effectively and appropriately in a culture of democracy

#### Values

#### Valuing human dignity and human rights

This value is based on the general belief that every human being is of equal worth, has equal dignity, is entitled to equal respect, and is entitled to the same set of human rights and fundamental freedoms, and ought to be treated accordingly.

#### Valuing cultural diversity

This value is based on the general belief that other cultural affiliations, cultural variability and diversity, and pluralism of perspectives, views and practices ought to be positively regarded, appreciated and cherished.

#### Valuing democracy, justice, fairness, equality and the rule of law

This set of values is based on the general belief that societies ought to operate and be governed through democratic processes which respect the principles of justice, fairness, equality and the rule of law.

#### **Attitudes**

# Openness to cultural otherness and to other beliefs, world views and practices

Openness is an attitude towards people who are perceived to have different cultural affiliations from oneself or towards beliefs, world views and practices which differ from one's own. It involves sensitivity towards, curiosity about and willingness to engage with other people and other perspectives on the world.

#### Respect

Respect consists of positive regard and esteem for someone or something based on the judgment that they have intrinsic importance, worth or value. Having respect for other people who are perceived to have different cultural affiliations or different beliefs, opinions or practices from one's own is vital for effective intercultural dialogue and a culture of democracy.

#### Civic-mindedness

Civic-mindedness is an attitude towards a community or social group to which one belongs that is larger than one's immediate circle of family and friends. It involves a sense of belonging to that community, an awareness of other people in the community, an awareness of the effects of one's actions on those people,

solidarity with other members of the community and a sense of civic duty towards the community.

#### Responsibility

Responsibility is an attitude towards one's own actions. It involves being reflective about one's actions, forming intentions about how to act in a morally appropriate way, conscientiously performing those actions and holding oneself accountable for the outcomes of those actions.

#### **Self-efficacy**

Self-efficacy is an attitude towards the self. It involves a positive belief in one's own ability to undertake the actions that are required to achieve particular goals, and confidence that one can understand issues, select appropriate methods for accomplishing tasks, navigate obstacles successfully and make a difference in the world.

#### **Tolerance of ambiguity**

Tolerance of ambiguity is an attitude towards situations which are uncertain and subject to multiple conflicting interpretations. It involves evaluating these kinds of situations positively and dealing with them constructively.

#### Skills

#### **Autonomous learning skills**

Autonomous learning skills are the skills required to pursue, organise and evaluate one's own learning in accordance with one's own needs, in a self-directed manner, without being prompted by others.

#### **Analytical and critical thinking skills**

Analytical and critical thinking skills are the skills required to analyse, evaluate and make judgments about materials of any kind (e.g. texts, arguments, interpretations, issues, events, experiences, etc.) in a systematic and logical manner.

#### **Skills of listening and observing**

Skills of listening and observing are the skills required to notice and understand what is being said and how it is being said, and to notice and understand other people's non-verbal behaviour.

#### **Empathy**

Empathy is the set of skills required to understand and relate to other people's thoughts, beliefs and feelings, and to see the world from other people's perspectives.

#### Flexibility and adaptability

Flexibility and adaptability are the skills required to adjust and regulate one's thoughts, feelings or behaviours so that one can respond effectively and appropriately to new contexts and situations.

#### Linguistic, communicative and plurilingual skills

Linguistic, communicative and plurilingual skills are the skills required to communicate effectively and appropriately with people who speak the same or another language, and to act as a mediator between speakers of different languages.

#### **Co-operation skills**

Co-operation skills are the skills required to participate successfully with others in shared activities, tasks and ventures and to encourage others to co-operate so that group goals may be achieved.

#### **Conflict-resolution skills**

Conflict-resolution skills are the skills required to address, manage and resolve conflicts in a peaceful way by guiding conflicting parties towards optimal solutions that are acceptable to all parties.

#### **Knowledge and critical understanding**

#### Knowledge and critical understanding of the self

This includes knowledge and critical understanding of one's own thoughts, beliefs, feelings and motivations, and of one's own cultural affiliations and perspective on the world.

# Knowledge and critical understanding of language and communication

This includes knowledge and critical understanding of the socially appropriate verbal and non-verbal communicative conventions that operate in the language(s) which one speaks, of the effects that different communication styles can have on other people, and of how every language expresses culturally shared meanings in a unique way.

#### Knowledge and critical understanding of the world

This includes a large and complex body of knowledge and critical understanding in a variety of areas including politics, law, human rights, culture, cultures, religions, history, media, economies, the environment and sustainability.

### 1. Introduction

his document proposes a conceptual model of the competences which enable citizens to participate effectively in a culture of democracy. The purpose of the model is to describe the competences which need to be acquired by learners if they are to become effective participatory citizens and live peacefully together with others as equals in culturally diverse democratic societies. It is intended that the model outlined in this document will be used to inform educational decision making and planning, helping educational systems to be harnessed for the preparation of learners for life as competent democratic citizens.<sup>1</sup>

The term "culture of democracy" rather than "democracy" is used in the present context to emphasise the fact that, while democracy cannot exist without democratic institutions and laws, such institutions and laws cannot work in practice unless they are grounded in a culture of democracy, that is, in democratic values, attitudes and practices. Among other things, these include a commitment to the rule of law and human rights, a commitment to the public sphere, a conviction that conflicts must be resolved peacefully, acknowledgement of and respect for diversity, a willingness to express one's own opinions, a willingness to listen to the opinions of others, a commitment to decisions being made by majorities, a commitment to the protection of minorities and their rights, and a willingness to engage in dialogue across cultural divides.

The model which is described in this document regards intercultural dialogue as being of central importance to democratic processes within culturally diverse societies. A fundamental principle of democracy is that the people who are affected by political decisions should be able to express their views when those decisions are being made and that decision makers should pay attention to those views. Intercultural dialogue is the single most important means through which citizens can express their opinions, aspirations, concerns and needs to those who have different cultural affiliations from themselves. This means that, in culturally diverse societies, intercultural dialogue is crucial for democratic discussion, debate and deliberation, and for enabling all citizens to contribute to political decision making on an equal footing. Likewise, democratic attitudes are crucial for intercultural dialogue because it is only when individuals regard each other as democratic equals that truly respectful communication and dialogue may take place between them. A culture of democracy and intercultural dialogue are inherently interdependent in culturally diverse societies.

<sup>1.</sup> In this document, the term "citizens" is used to denote all individuals who are affected by democratic decision making and who can engage with democratic processes and institutions (rather than to denote only those who hold legal citizenship and the passport of a particular state).

As the Council of Europe's White Paper on intercultural dialogue (2008)² notes, the competences which citizens need to acquire if they are to participate effectively in a culture of democracy are not acquired automatically but instead need to be learned and practised. Education has a key role to play in this regard. Education has many purposes, including preparing individuals for the labour market, supporting personal development and providing a broad advanced knowledge base within society. However, in addition, education has a vital role to play in preparing individuals for life as active democratic citizens, and education is in a unique position to guide and support learners in acquiring the competences which they require to participate effectively in democratic processes and intercultural dialogue.

An education system which equips people with such competences empowers them, endowing them with the capacities which they need to become active participants in democratic processes, in intercultural dialogue and in society more generally. It also endows them with the ability to function as autonomous social agents capable of choosing and pursuing their own goals in life. The current competence model has been developed to assist educational planning towards this goal of empowering all learners.<sup>3</sup>

<sup>2.</sup> Council of Europe (2008). White Paper on intercultural dialogue: "Living together as equals in dignity". Strasbourg: Committee of Ministers, Council of Europe. Available at www.coe.int/t/dg4/intercultural/source/white%20paper\_final\_revised\_en.pdf, accessed 17 February 2016.

<sup>3.</sup> It should be noted that the current framework is intended to apply not only to learners in mainstream education but also to learners with special educational needs. Learners with special educational needs (who may have physical, learning, emotional, behavioural or sensory disabilities or impairments, and may or may not be in mainstream education) are entitled to exactly the same human rights and fundamental freedoms as all other learners, including the right to education. The use of the current framework to support planning for such learners is vital for their empowerment to enable them to participate in democratic processes and intercultural dialogue to their fullest potential.

# 2. A background assumption: competences are necessary but not sufficient

efore outlining the model, it is important to clarify two background assumptions that underlie the current approach. The first of these is that, while it is necessary for citizens to acquire a range of competences in order to participate effectively in democratic processes, these competences are not sufficient for such participation to occur. This is for the following reasons.

First, a culture of democracy requires, in addition to competent citizens, suitable political and legal structures and procedures to support citizens' exercise of their competences. This is because institutional structures and procedures, and the opportunities for active engagement which they make available or deny to citizens, can serve as significant enablers or inhibitors of the democratic and intercultural actions that citizens are able to undertake. To give a simple example, if a country denies the right to vote in national elections to first-generation migrants prior to their naturalisation, then no matter how democratically competent a firstgeneration migrant might be, they will be unable to exercise that competence in national elections unless they have been naturalised. A more complex example: citizens' opportunities for democratic activities and participation are limited if there are no or few institutional consultative channels or bodies through which citizens can communicate their views to politicians and policy makers. Where this occurs, citizens will need to pursue alternative forms of democratic action if they wish to make their voices heard. A third example: if there are no institutional arrangements or structures to support intercultural dialogue, then citizens are less likely to engage in such dialogue. However, if governments take active steps to establish or facilitate an abundance of places and spaces for dialogue (e.g. cultural and social centres, youth clubs, education centres, other leisure facilities or virtual spaces), and promote and encourage the use of these facilities for intercultural activities, then citizens are more likely to engage in intercultural dialogue. In other words, depending upon their configuration, institutional arrangements can enable, channel, constrain or inhibit the ways in which citizens exercise their democratic and intercultural competences. The exercise of competences is not solely dependent on whether or not they have been acquired. Thus, while democratic institutions are not self-sustaining (without an accompanying culture of democracy), it is also the case that a culture of democracy is not self-sustaining (in the absence of appropriate institutions).

The acquisition of democratic and intercultural competences may also not be sufficient to enable citizens to participate in democratic processes and intercultural dialogue for another reason. The presence of systematic patterns of disadvantage and discrimination, and differentials in the allocation of resources within societies, may effectively disempower many people from participating on an equal footing. irrespective of their levels of competence (e.g. by limiting their access to sources of information, or their access to the time or the financial resources which are needed to participate). These inequalities and disadvantages are often further compounded by disparities of power and by institutional biases which lead to democratic and intercultural settings and opportunities being dominated by those who occupy positions of privilege. Systematic marginalisation and exclusion from democratic processes and intercultural exchanges can lead to citizens' civic disengagement and alienation. For all of these reasons, special measures need to be adopted to ensure that members of disadvantaged groups enjoy genuine equality of condition. In other words, it is not sufficient only to equip citizens with the competences that are specified by the current model. It is also necessary to adopt measures to tackle inequalities and structural disadvantages.

Consequently, the current approach presupposes that the competences which are described in this document are necessary for participation in democratic processes and intercultural dialogue, but are not sufficient to ensure such participation. Although this document focuses only on the competences which citizens require, the need for appropriate institutional structures and for action on inequalities and structural disadvantages should be borne in mind throughout.

# 3. A further background assumption: all cultures are internally heterogeneous, contested, dynamic and constantly evolving

second background assumption underlying the current approach concerns the concept of culture that it employs. "Culture" is a difficult term to define, largely because cultural groups are always internally heterogeneous and embrace a range of diverse practices and norms that are often disputed, change over time and are enacted by individuals in personalised ways. That said, any given culture may be construed as having three main aspects: the material resources that are used by members of the group (e.g. tools, foods, clothing), the socially shared resources of the group (e.g. the language, religion, rules of social conduct) and the subjective resources that are used by individual group members (e.g. the values, attitudes, beliefs and practices which group members commonly use as a frame of reference for making sense of and relating to the world). The culture of the group is a composite formed from all three aspects – it consists of a network of material, social and subjective resources. The total set of resources is distributed across the entire group, but each individual member of the group appropriates and uses only a subset of the total set of cultural resources potentially available to them.

Defining "culture" in this way means that groups of any size can have their own distinctive cultures. This includes nations, ethnic groups, religious groups, cities, neighbourhoods, work organisations, occupational groups, sexual orientation groups, disability groups, generational groups and families. For this reason, all people belong simultaneously to and identify with many different groups and their associated cultures.

There is usually considerable variability within cultural groups because the resources which are perceived to be associated with membership of the group are often resisted, challenged or rejected by different individuals and subgroups within it. In addition, even the boundaries of the group itself, and who is perceived to be within the group and who is perceived to be outside the group, may be disputed by different group members – cultural group boundaries are often very fuzzy.

This internal variability and contestation of cultures is, in part, a consequence of the fact that all people belong to multiple groups and their cultures but participate in different constellations of cultures, so that the ways in which they relate to any one culture depend, at least in part, on the points of view which are present in the other cultures in which they also participate. In other words, cultural affiliations intersect in such a way that each person occupies a unique cultural positioning. In addition, the meanings and feelings which people attach to particular cultures are personalised as a consequence of their own life histories, personal experiences and individual personalities.

Cultural affiliations are fluid and dynamic, with the subjective salience of social and cultural identities fluctuating as individuals move from one situation to another, with different affiliations – or different clusters of intersecting affiliations – being highlighted depending on the particular social context encountered. Fluctuations in the salience of cultural affiliations are also linked to shifts in people's interests, needs, goals and expectations as they move across situations and through time. Furthermore, all groups and their cultures are dynamic and change over time as a result of political, economic and historical events and developments, and as a result of interactions with and influences from the cultures of other groups. They also change over time because of their members' internal contestation of the meanings, norms, values and practices of the group.

This underlying concept of culture which was used to develop the current model has implications for the concept of "intercultural". If we all participate in multiple cultures, but we each participate in a unique constellation of cultures, then every interpersonal situation is potentially an intercultural situation. Often, when we encounter other people, we respond to them as individuals who have a range of physical, social and psychological attributes which serve to distinguish them from other people. However, sometimes we respond to them instead in terms of their cultural affiliations, and when this occurs we group them together with other people who share these affiliations with them. There are several factors which prompt us to shift our frame of reference from the individual and interpersonal to the intercultural. These include, inter alia, the presence of salient cultural emblems or practices that elicit or invoke the cultural category in the mind of the perceiver, the frequent use of cultural categories to think about other people so that these categories are primed and readily accessed when interacting with others, and the usefulness of a cultural category in helping to understand why another person is behaving in the way that they are.

Thus, intercultural situations arise when an individual perceives another person (or group of people) as being culturally different from themselves. When other people are perceived as members of a social group and its culture rather than as individuals, then the self is also usually categorised – and may present itself – as a cultural group member rather than in purely individual terms. Intercultural situations, identified in this way, may involve people from different countries, people from different regional, linguistic, ethnic or faith groups, or people who differ from each other because of their lifestyle, gender, age or generation, social class, education, occupation, level of religious observance, sexual orientation, etc. From this perspective, intercultural dialogue may be defined as "an open exchange of views, on the basis of mutual

understanding and respect, between individuals or groups who perceive themselves as having different cultural affiliations from each other".

There is good scientific evidence that intercultural dialogue fosters constructive engagement across perceived cultural divides, reduces intolerance, prejudice and stereotyping, enhances the cohesion of democratic societies and helps to resolve conflicts. That said, intercultural dialogue can be a difficult process. This is particularly the case when the participants perceive each other as representatives of cultures that have an adversarial relationship with one another (e.g. as a consequence of past or present armed conflict) or when a participant believes that their own cultural group has experienced significant harm (e.g. blatant discrimination, material exploitation or genocide) at the hands of another group to which they perceive their interlocutor as belonging. Under such circumstances, intercultural dialogue can be extremely difficult, requiring a high level of intercultural competence and very considerable emotional and social sensitivity, commitment, perseverance and courage.

In short, the present approach assumes that cultures are internally heterogeneous, contested, dynamic and constantly changing, and that intercultural situations arise due to the perception that there are cultural differences between people. Hence, the current model of competences makes frequent reference to "people who are perceived to have different cultural affiliations from oneself" (rather than to, for example, "people from other cultures"). Intercultural dialogue is construed as an open exchange of views between individuals or groups who perceive themselves as having different cultural affiliations from each other. Intercultural dialogue is extremely important for fostering constructive engagement across cultural divides and for enhancing the cohesion of democratic societies, although full openness may be very difficult to achieve under some circumstances.

# 4. The concept of competence employed in the current model

he term "competence" can be used in many ways, including its casual everyday use as a synonym for "ability", its more technical use within vocational education and training, and its use to denote the ability to meet complex demands within a given context. For the purposes of the current model, the term "competence" is defined as "the ability to mobilise and deploy relevant values, attitudes, skills, knowledge and/or understanding in order to respond appropriately and effectively to the demands, challenges and opportunities that are presented by a given type of context". Democratic situations are one such type of context. Thus, democratic competence is the ability to mobilise and deploy relevant psychological resources (i.e. values, attitudes, skills, knowledge and/or understanding) in order to respond appropriately and effectively to the demands, challenges and opportunities presented by democratic situations. Likewise, intercultural competence is the ability to mobilise and deploy relevant psychological resources in order to respond appropriately and effectively to the demands, challenges and opportunities presented by intercultural situations. In the case of citizens who live within culturally diverse democratic societies, intercultural competence is construed as being an integral aspect of democratic competence (for the reasons given in Chapter 1).

It is important to note that democratic and intercultural situations occur not only in the physical world but also in the digital online world. In other words, democratic debates and deliberations and intercultural encounters and interactions take place not only through face-to-face exchanges, traditional print and broadcast media, letters, petitions, etc., but also through computer-mediated communications, for example, through online social networks, forums, blogs, e-petitions and e-mails. The competence model which is described in the present document applies not only to democratic and intercultural situations in the physical world but also to such situations in the digital world. For this reason, the model has relevance not only to education for democratic citizenship, human rights education and intercultural education but also to digital citizenship education.

The present model treats competence as a dynamic process. This is because competence involves the selection, activation, organisation and co-ordination of relevant psychological resources which are then applied through behaviour in such a way that the individual adapts appropriately and effectively to a given situation. Appropriate and effective adaptation involves the constant monitoring of the results of behaviour and of the situation, and the adjustment and modification of behaviour (which may entail the mobilisation of further psychological resources) if this is required to meet the shifting needs and demands of the situation. In other words, a competent individual mobilises and deploys psychological resources in a dynamic manner according to situational contingencies.

In addition to this global and holistic use of the term "competence" (in the singular), the term "competences" (in the plural) is used in the current account to refer to the specific individual resources (i.e. the specific values, attitudes, skills, knowledge and understanding) that are mobilised and deployed in the production of competent behaviour. Hence, on the present account, competence consists of the selection, activation and organisation of competences and the application of these competences in a co-ordinated, adaptive and dynamic manner to concrete situations.

It should be noted that, according to the current model, competences include not only skills, knowledge and understanding but also values and attitudes. Values and attitudes are regarded as essential for behaving appropriately and effectively in democratic and intercultural situations.

However, dispositions are excluded from the set of competences specified by the model. Dispositions are instead treated as being implicit in the definition of competence which underpins the entire model – that is, competence as the mobilisation and deployment of competences through behaviour. If competences are not mobilised and deployed (i.e. if there is no disposition to use them in behaviour), then an individual cannot be deemed to be competent. In other words, having the disposition to use one's competences in behaviour is intrinsic to the very notion of competence – there is no competence without this disposition.

In real-life situations, competences are rarely mobilised and used individually. Instead, competent behaviour invariably involves the activation and application of an entire cluster of competences. Depending on the situation, and the specific demands, challenges and opportunities which that situation presents, and also the specific needs and goals of the individual within that situation, different subsets of competences will need to be activated and deployed.

One example of how an entire cluster of competences has to be mobilised is provided by intercultural dialogue. Such dialogue initially requires an attitude of openness towards another person who is perceived to have cultural affiliations that differ from one's own. It may also require the ability to overcome anxieties or insecurities about meeting and interacting with someone with whom one feels one has little in common. However, once the dialogue commences, close listening skills and linguistic and communicative skills need to be mobilised and deployed to ensure that miscommunications do not occur and to ensure that the content of the dialogue remains sensitive to the communicative needs and cultural norms of the other person. Empathy is also likely to be required, along with analytical thinking skills, to

facilitate comprehension of the other person's point of view, especially when this is not immediately apparent from what he or she says. It may emerge during the course of the dialogue that there are irreconcilable differences in points of view between the self and the other. If this is the case, then tolerance of ambiguity will need to be deployed and the lack of a clear-cut resolution accepted. Thus, effective and appropriate behaviour in the context of intercultural dialogue requires the mobilisation, orchestration and sensitive application of a very large range of competences.

A further example is provided when a citizen takes a principled stand against hate speech that is being directed at refugees or migrants on the Internet. Such a stand is likely to be initiated through the activation of human dignity as a fundamental value and to be sustained through the activation of an attitude of civic-mindedness and a sense of responsibility. To challenge the contents of the hate speech, analytical and critical thinking skills will need to be applied. In addition, the formulation of an appropriate response requires knowledge of human rights as well as communicative skills in order to ensure that the stand that is taken is expressed appropriately and is targeted effectively at its intended audience(s). In addition, knowledge and understanding of digital media will need to be drawn upon to ensure that the response is posted in an appropriate manner and its impact maximised. Thus, effective behaviour in response to online hate speech also requires the mobilisation and orchestration of a wide range of competences.

A third example is provided by participation in political debate. In order to function effectively in debate, communications need to be adapted to both the medium of expression (e.g. speech, writing) and the intended audience. In addition, one needs to have a critical understanding of freedom of expression and its limits, and, in cases where communications involve people who are perceived to have different cultural affiliations from oneself, an understanding of cultural appropriateness. Political debate also requires knowledge and understanding of politics, and the ability to critique the views of others and to evaluate the arguments which they deploy during the course of the debate. Political debate therefore requires all of the following competences: linguistic and communicative skills, knowledge and understanding of communication, knowledge and understanding of cultural norms, knowledge and understanding of politics, analytical and critical thinking skills, and the ability to adapt one's arguments appropriately as the debate proceeds. Thus, once again, several competences have to be mobilised, co-ordinated and deployed appropriately and sensitively according to the specific contingencies that arise as the debate proceeds.

In summary, democratically and interculturally competent behaviour is viewed by the present model as arising from a dynamic and adaptive process in which an individual responds appropriately and effectively to the constantly shifting demands, challenges and opportunities that are presented by democratic and intercultural situations. This is achieved through the flexible mobilisation, orchestration and deployment of varying subsets of psychological resources, drawn selectively from the individual's full repertoire of values, attitudes, skills, knowledge and understanding.

# 5. The working method used to identify the competences to be included in the model

n important feature of the contents of the current competence model is that it has not been designed from scratch. Instead, it is grounded in an analysis of existing conceptual schemes of democratic competence and intercultural competence. Many such schemes have been formulated by the Council of Europe over the years, as well as by other international bodies (e.g. UNESCO, OECD, European Parliament) and by national governments and ministries of education. In addition, a large number of democratic and intercultural competence schemes have been formulated by academic researchers. These various schemes differ considerably in both structure and content. The proliferation of diverse models presents a dilemma to educational planners and policy makers who wish to find an authoritative model upon which to base their work. The model proposed in the present document is an attempt to provide an account which captures, synthesises and builds upon the optimal features of existing competence schemes.

#### 5.1. Audit of existing competence schemes

The method used to identify the psychological resources to be included in the current model began with an audit of existing schemes of democratic competence and intercultural competence. For the purposes of the audit, democratic and intercultural competence schemes were defined as being those schemes that identify and describe relevant sets of values, attitudes, skills, knowledge and/or understanding that need to be mobilised and deployed in order to respond appropriately and effectively to the demands, challenges and opportunities presented by democratic and intercultural situations, respectively.

The audit only included competence schemes defined in this way within its purview. There are other types of framework that have been formulated to explain either democratic or intercultural behaviour but which do not have as their primary goal the identification and description of relevant sets of values, attitudes, skills, knowledge and/or understanding. These other frameworks are instead aimed at explaining either: (i) the social and/or psychological processes through which democratic or intercultural behaviour is generated; (ii) the factors that influence the democratic or intercultural behaviour of individuals; or (iii) the developmental sequence or stages through which individuals acquire democratic or intercultural behaviour. These process models, influence models and developmental models were excluded from the audit, unless they also provide an explicit description of relevant sets of values, attitudes, skills, knowledge and/or understanding.

The audit identified and collated a total number of 101 competence schemes which met the definition given above. References for these 101 schemes are given in Appendix A. There are several points to note about these various schemes.

First, some schemes are articulated as explicit lists of competences, whereas other schemes are articulated in much lengthier texts which required some work to extract the descriptions of specific competences from them.

Second, the schemes that were audited vary in terms of the methods that were used by their authors to construct them. For example, some are based on systematic reviews of the research literature; some are derived statistically from survey and other empirical data; some are based on information received from specialist informants such as teachers, academics, researchers, graduates or employers; some are based on reviews of previous conceptual schemes; some are based on the opinion of an expert group; and some are based on the opinion of an individual expert.

Third, the schemes are not always independent of one another. For example, some authors have been especially prolific and have produced multiple schemes that often display considerable overlap, and some authors have adopted and modified in only relatively minor ways the competence schemes proposed by other authors.

Fourth, while the majority of schemes provide descriptions of the competences required by citizens in general, a few of the schemes provide descriptions of the more specialised competences that are required by particular professional groups such as teachers or psychological counsellors.

Fifth, and perhaps most problematically, there are widespread mismatches across these various schemes in terms of: (i) the number of competences which they contain; (ii) the level of generality at which these competences are described; (iii) the particular competences which are included; and (iv) the ways in which the competences are grouped and classified.

#### **5.2.** Decomposition of existing competence schemes

In the next step of the analysis, the 101 schemes were decomposed to identify all the individual competences which they contained, and these competences were then grouped into cognate sets. This process led to the identification of 55 possible competences that are relevant to a culture of democracy across the 101 schemes. These 55 competences are listed in Appendix B. It should be noted that some competences were present across a relatively large number of schemes, whereas other competences were specific to just one or a very small number of schemes.

Not all of the competences could be placed into particular sets with full confidence. The difficulties in grouping the competences into sets occurred for several reasons. For example, some competences, in the form in which they are specified within their conceptual schemes, are inherently ambiguous or vague. Other competences are formulated within their schemes in a way which combines what would appear to be two or more distinct competences, with the result that it is difficult to know what the primary focus is intended to be. In cases where this kind of "competence" was detected, that particular "competence" was included multiple times in the sets listed in Appendix B under all of the headings that might possibly apply. However, in a few cases, exactly the same combinations of what could be construed as two or more distinct competences were found to be present in a large number of the audited schemes. Because of the apparent consensus about these particular combinations, some of the headings shown in Appendix B provisionally maintained these more complex units.

# **5.3.** The identification of competences for inclusion in the model

To assist in reducing this list of 55 possible competences to a more manageable and practical length and to systematise the contents of the list, a set of principled criteria was articulated for including a competence in the new model. It was decided that, for inclusion in the model, competences needed to match seven criteria.

#### (i) Conceptually clear

Only competences which were conceptually clear were included in the model. Competences which were vague, imprecise or of indeterminate meaning were excluded.

#### (ii) Formulated at a general rather than at a specific level

A second principle was that the competences should be captured at a general rather than specific level of description, so that the model would be as comprehensive as possible without being exhaustive on specific details. This would enable users to adapt the general description as appropriate to their own particular contexts and needs.

#### (iii) Not tied to a particular professional role

A corollary to the preceding point was that the competences in the model should not be tied specifically to one particular professional role (such as teacher or psychological counsellor). Nevertheless, the option should remain open for users to adapt the more general model to specific professional groups as appropriate, if this is required.

#### (iv) Not idiosyncratic to just one or a very few of the audited competence schemes

A fourth principle was that the competences in the model should not be idiosyncratic to just one or a very few of the audited competence schemes. Instead, the competences to be extracted for inclusion in the model should be those upon which there was a substantial degree of consensus across the audited schemes.

#### (v) Not the behaviours through which competence is exhibited

It was also decided to restrict the competences in the model to internal psychological resources (i.e. to values, attitudes, skills, knowledge and understanding) and to exclude the behaviours through which these resources are exhibited. Behaviours themselves are the external outcomes of applying competences to meet the demands, challenges and opportunities presented by specific situations. Every competence has behavioural expressions, and all competent behaviours are a product of deploying one or more underlying competences. Behaviours are not separate competences in themselves, but are instead a means to assess whether or not individual competences have been acquired.<sup>4</sup>

#### (vi) Conceptually distinct from other competences

A further principle was that the competences included in the model should be conceptually distinct from one another. This principle was adopted to avoid problems in the development of descriptors for each competence, by minimising the likelihood of the same descriptors applying to multiple competences. In practice, this criterion was difficult to maintain consistently. It is therefore acknowledged that there are occasional cases of partial overlap between the competences which were eventually incorporated into the model (e.g. between co-operation skills and conflict-resolution skills). Conceptual distinctiveness nevertheless proved to be a useful general heuristic for guiding the development of the model.

#### (vii) Not generic competences

Finally, it was also decided that the competences to be included in the model should not include generic competences that apply across a wide range of domains of activity (in the way that, for example, literacy, numeracy and ICT skills apply across many domains). It was judged that this was necessary to enable the model to retain its core focus and to prevent it from overextending itself into an all-encompassing framework. However, once again, this criterion was difficult to maintain consistently in practice, and it is acknowledged that some of the competences that were extracted for inclusion in the model do have relevance to other domains in addition to democratic culture. These particular competences (e.g. analytical and critical thinking skills and autonomous learning skills) were included because they were deemed to be essential to a culture of democracy and it was judged that the model would clearly be incomplete if they were to be excluded.

<sup>4.</sup> Behaviours are instead incorporated into the current framework through the descriptors. Descriptors have been developed for all of the competences specified in the model, and have been formulated using the language of learning outcomes (i.e. in terms of observable and assessable behaviours). Thus, the descriptors are able to serve as indicators of whether or not particular competences have been mastered by learners.

In addition to the preceding set of principled criteria, a set of pragmatic considerations was adopted to assist in reducing the number of competences for incorporation into the model. These pragmatic considerations were the same as those that were used to develop the Common European framework of reference for languages. The considerations were that the model should be:

- multi-purpose: usable for the full variety of purposes required in educational planning and provision, including curriculum design, the development of programmes and methods of instruction, the development of methods of assessment, matching competences to levels of education, and matching competences (and possibly proficiency levels) to qualifications frameworks;
- ▶ **flexible**: adaptable for use in different circumstances, cultural settings and a variety of educational contexts;
- open: capable of further extension and refinement according to the needs of its users;
- **dynamic**: capable of being modified in response to feedback from its users;
- ▶ **non-dogmatic**: not exclusively attached to any one of a number of competing educational or social-scientific theories or practices;
- ▶ user-friendly: presented in a form that is readily understandable and usable by those to whom it is addressed, in particular educational practitioners and policy makers.<sup>6</sup>

Finally, because a core goal of the project was to construct a model for use in educational planning, it was also decided to ensure that all of the competences that were included in the model should be teachable, learnable and assessable (through either self-assessment or assessment by others). In practice, all potential competences that were identified for inclusion in the model were judged to meet these three criteria.

Applying the above criteria and considerations to the set of 55 possible competences listed in Appendix B led to the identification of 20 competences for inclusion in the model. These were as follows:

#### **Values**

- ► Valuing human dignity and human rights
- Valuing cultural diversity
- ▶ Valuing democracy, justice, fairness, equality and the rule of law

#### **Attitudes**

- ▶ Openness to cultural otherness and to other beliefs, world views and practices
- ► Respect
- 5. Council of Europe (2001). Common European framework of reference for languages: learning, teaching, assessment (CEFR), Cambridge University Press, Cambridge. Available at www.coe.int/t/dg4/linguistic/source/framework\_en.pdf, accessed 17 February 2016.
- 6. It is recognised that the current document, the aim of which is to describe the assumptions, rationale and technical details of the competence model, and the methods that were used to develop it, is not in a suitable format for use by practitioners and policy makers. More accessible documents describing the model will be produced at a later phase of the project.

- Civic-mindedness
- ► Responsibility
- ► Self-efficacy
- ► Tolerance of ambiguity

#### Skills

- ► Autonomous learning skills
- Analytical and critical thinking skills
- Skills of listening and observing
- ► Empathy
- ► Flexibility and adaptability
- ► Linguistic, communicative and plurilingual skills
- ► Co-operation skills
- Conflict-resolution skills

#### Knowledge and critical understanding

- ▶ Knowledge and critical understanding of the self
- ▶ Knowledge and critical understanding of language and communication
- ► Knowledge and critical understanding of the world (including politics, law, human rights, culture, cultures, religions, history, media, economies, the environment and sustainability)

The 20 competences contained in the model are summarised diagrammatically on page 35.

The 20 competences were then subjected to a further process of conceptual refinement and clarification in their formulation. In generating the descriptive details for each of the competences, constant reference was made back to the descriptions of the competences in the original 101 competence schemes from which they had been derived, to ensure that the most significant aspects of each competence as described in the audited schemes were being captured by the model. Many other sources of information were also drawn upon in formulating the descriptive details of the model and in writing the current document. Some of these additional sources are included as suggestions for further reading in Appendix C.

A first draft of the current document was then circulated in February 2015 in an international consultation exercise which involved academic experts, educational practitioners and policy makers, including experts nominated by the education ministries of the member states of the Council of Europe. These stakeholders were invited to provide feedback and comments, particularly concerning the conceptual soundness of the model, whether there were any significant omissions from the model, and the clarity of the text. The model was also presented at various conferences, workshops and meetings attended by academic experts, educational practitioners and policy makers during the first six months of 2015, where additional feedback and comments were gathered. It is noteworthy that, overall, the process of consultation

resulted in the strong endorsement of the model by the stakeholders. However, many useful suggestions were also received concerning technical details of the model and the precise wordings and mode of presentation used in the document. This feedback was used to fine-tune details of the model and to guide the writing of the present document.

The following text describes the competence model that was constructed through the use of this working method.

## 6. The model

ithin the context of democratic culture, an individual is deemed to be acting competently when he or she meets the demands, challenges and opportunities that are presented by democratic and intercultural situations appropriately and effectively by mobilising and deploying some or all of the following competences. The competences fall into four broad categories: values, attitudes, skills, and knowledge and critical understanding.

#### The 20 competences included in the model

#### **Values**

- Valuing human dignity and human rights
- Valuing cultural diversity
- Valuing democracy, justice, fairness, equality and the rule of law

#### **Attitudes**

- Openness to cultural otherness and to other beliefs, world views and practices
- Respect
- Civic-mindedness
- Responsibility
- Self-efficacy
- Tolerance of ambiguity

#### **Competence**

- Autonomous learning skills
- Analytical and critical thinking skills
- Skills of listening and observing
- Empathy
- Flexibility and adaptability
- Linguistic, communicative and plurilingual skills
- Co-operation skills
- Conflict-resolution skills

#### **Skills**

- Knowledge and critical understanding of the self
- Knowledge and critical understanding of language and communication
- Knowledge and critical understanding of the world: politics, law, human rights, culture, cultures, religions, history, media, economies, environment, sustainability

Knowledge and critical understanding

# 6.1. Values

Values are general beliefs that individuals hold about the desirable goals that should be striven for in life. They motivate action and they also serve as guiding principles for deciding how to act. Values transcend specific actions and contexts, and they have a normative prescriptive quality about what ought to be done or thought across many different situations. Values offer standards or criteria for: evaluating actions, both one's own and those of other people; justifying opinions, attitudes and behaviours; deciding between alternatives; planning behaviour; and attempting to influence others.

Readers familiar with existing competence schemes may be surprised by the appearance of values as a distinct type of competence in the current model. However, it is important to bear in mind that the term "competence" is not being used here in its casual everyday sense as a synonym of "ability", but in a more technical sense to refer to the psychological resources (such as attitudes, skills and knowledge) which need to be mobilised and deployed to meet the demands and challenges of democratic and intercultural situations. Values are included in the model for two reasons. First, values do in fact appear (although often only implicitly) in many of the previous competence schemes which were audited. Thus, their omission from the current model would have meant that the model was only partially capturing the contents of those previous schemes. Values are not always obvious in previous schemes because they are usually included under the heading of attitudes (rather than under their own distinct heading). By contrast, the current model draws a clear conceptual distinction between values and attitudes, with only the former being characterised by their normative prescriptive quality. Second, values are essential in the context of conceptualising the competences which enable participation in a culture of democracy. This is because, without a specification of the particular values that underpin these competences, they would not be democratic competences but would instead be more general political competences which could be used in the service of many other kinds of political order, including anti-democratic orders. For example, one could be a responsible, self-efficacious and politically well-informed citizen within a totalitarian dictatorship if a different set of values were to be employed as the foundation for one's judgments, decisions and actions. Thus, the values which the current model contains lie at the very heart of democratic competence, and are essential for the characterisation of that competence.

There are three sets of values that are crucial for participating in a culture of democracy, as follows.

# Valuing human dignity and human rights

This first set of values is based on the general belief that every individual human being is of equal worth, has equal dignity, is entitled to equal respect, and is entitled to the same set of human rights and fundamental freedoms, and ought to be treated accordingly. This belief assumes: that human rights are universal, inalienable and indivisible and apply to everyone without distinction; that human rights provide a minimum set of protections that are essential for human beings to live a life of

dignity; and that human rights provide an essential foundation for freedom, equality, justice and peace in the world. This set of values therefore involves:

- 1. recognition that all people share a common humanity and have equal dignity irrespective of their particular cultural affiliations, status, abilities or circumstances;
- 2. recognition of the universal, inalienable and indivisible nature of human rights;
- 3. recognition that human rights should always be promoted, respected and protected;
- 4. recognition that fundamental freedoms should always be defended unless they undermine or violate the human rights of others;
- 5. recognition that human rights provide the foundation for living together as equals in society and for freedom, justice and peace in the world.

# **Valuing cultural diversity**

The second set of values is based on the general belief that other cultural affiliations, cultural variability and diversity, and pluralism of perspectives, views and practices ought to be positively regarded, appreciated and cherished. This belief assumes: that cultural diversity is an asset for society; that people can learn and benefit from other people's diverse perspectives; that cultural diversity should be promoted and protected; that people should be encouraged to interact with one another irrespective of their perceived cultural differences; and that intercultural dialogue should be used to develop a democratic culture of living together as equals in society.

Note that there is a tension between valuing human rights and valuing cultural diversity. In a society which has adopted human rights as its primary value foundation, valuing cultural diversity will have certain limits. These limits are set by the need to promote, respect and protect the human rights and freedoms of other people. Hence, it is assumed here that cultural diversity always ought to be valued unless it undermines the human rights and freedoms of others.

This second set of values therefore involves:

- recognition that cultural diversity and pluralism of opinions, world views and practices is an asset for society and provides an opportunity for the enrichment of all members of society;
- 2. recognition that all people have the right to be different and the right to choose their own perspectives, views, beliefs and opinions;
- 3. recognition that people should always respect the perspectives, views, beliefs and opinions of other people, unless these are directed at undermining the human rights and freedoms of others;
- 4. recognition that people should always respect the lifestyles and practices of other people, unless they undermine or violate the human rights and freedoms of others;
- 5. recognition that people should listen to and engage in dialogue with those who are perceived to be different from themselves.

# Valuing democracy, justice, fairness, equality and the rule of law

The third set of values is based on a cluster of beliefs about how societies ought to operate and be governed, including the beliefs that: all citizens ought to be able to participate equally (either directly or indirectly through elected representatives) in the procedures through which the laws that are used to regulate society are formulated and established; all citizens ought to engage actively with the democratic procedures which operate within their society (allowing that this might also mean not engaging on occasions for reasons of conscience or circumstance); while decisions ought to be made by majorities, the just and fair treatment of minorities of all kinds ought to be ensured; social justice, fairness and equality ought to operate at all levels of society; and the rule of law ought to prevail so that everyone in society is treated justly, fairly, impartially and equally in accordance with laws that are shared by all. This set of values therefore involves:

- support for democratic processes and procedures (while recognising that existing democratic procedures may not be optimal and that there may sometimes be a need to change or improve them through democratic means);
- recognition of the importance of active citizenship (while recognising that non-participation may sometimes be justified for reasons of conscience or circumstance);
- 3. recognition of the importance of citizen engagement with political decision making;
- 4. recognition of the need for the protection of civil liberties, including the civil liberties of people who hold minority views;
- 5. support for the peaceful resolution of conflicts and disputes;
- a sense of social justice and social responsibility for the just and fair treatment of all members of society, including equal opportunities for all irrespective of national origins, ethnicity, race, religion, language, age, sex, gender, political opinion, birth, social origin, property, disability, sexual orientation or other status;
- 7. Support for the rule of law and the equal and impartial treatment of all citizens under the law as a means of ensuring justice.

# 6.2. Attitudes

An attitude is the overall mental orientation which an individual adopts towards someone or something (e.g. a person, a group, an institution, an issue, an event, a symbol). Attitudes usually consist of four components: a belief or opinion about the object of the attitude, an emotion or feeling towards the object, an evaluation (either positive or negative) of the object, and a tendency to behave in a particular way towards that object.

Six attitudes that are important for a culture of democracy are as follows.

# Openness to cultural otherness and to other beliefs, world views and practices

Openness is an attitude either towards people who are perceived to have different cultural affiliations from oneself or towards world views, beliefs, values and practices that differ from one's own. The attitude of openness towards cultural otherness needs to be distinguished from the attitude of having an interest in collecting experiences of the "exotic" merely for one's own personal enjoyment or benefit. Openness instead involves:

- 1. sensitivity towards cultural diversity and to world views, beliefs, values and practices which differ from one's own;
- curiosity about, and interest in discovering and learning about, other cultural orientations and affiliations and other world views, beliefs, values and practices;
- 3. willingness to suspend judgment and disbelief of other people's world views, beliefs, values and practices, and willingness to question the "naturalness" of one's own world view, beliefs, values and practices;
- 4. emotional readiness to relate to others who are perceived to be different from oneself:
- 5. willingness to seek out or take up opportunities to engage, co-operate and interact with those who are perceived to have cultural affiliations that differ from one's own, in a relationship of equality.

# Respect

Respect is an attitude towards someone or something (e.g. a person, a belief, a symbol, a principle, a practice) where the object of that attitude is judged to have some kind of importance, worth or value which warrants positive regard and esteem.<sup>7</sup> Depending on the nature of the object that is respected, the respect may take on very different forms (cf. respect for a school rule v. respect for an elder's wisdom v. respect for nature).

<sup>7.</sup> Notice that respect is closely linked to values in two ways: a value may be an object of respect (i.e. a value may be respected) and it can also function as a foundation for respect (i.e. one can respect someone or something because they exemplify or put into practice a particular value).

One type of respect that is especially important in the context of a culture of democracy is the respect that is accorded to other people who are perceived to have different cultural affiliations or different beliefs, opinions or practices from one's own. Such respect assumes the intrinsic dignity and equality of all human beings and their inalienable human right to choose their own affiliations, beliefs, opinions or practices. Importantly, this type of respect does not require minimising or ignoring the actual differences that might exist between the self and the other, which can sometimes be significant and profound, nor does it require agreement with, adoption of or conversion to that which is respected. It is instead an attitude that involves the positive appreciation of the dignity and the right of the other person to hold those affiliations, beliefs, opinions or practices, while nevertheless recognising and acknowledging the differences which exist between the self and the other. An attitude of respect is required to facilitate both democratic interaction and intercultural dialogue with other people. However, it should be noted that limits do need to be placed on respect – for example, respect should not be accorded to the contents of beliefs and opinions, or to lifestyles and practices which undermine or violate the dignity, human rights or freedoms of others.8

The concept of respect reflects better than the concept of tolerance the attitude that is required for a culture of democracy. Tolerance may, in some contexts, convey the connotation of simply enduring or putting up with difference and a patronising stance of tolerating something that one would prefer not to endure. Tolerance may also sometimes be construed as involving an act of power which allows the existence of difference by merely tolerating it, and through this act of tolerance enhancing the power and authority of the tolerating individual. Respect is a less ambiguous concept than tolerance, being based on recognition of the dignity, rights and freedoms of the other and a relationship of equality between the self and the other.

# Respect therefore involves:

- 1. positive regard and esteem for someone or something based on the judgment that they have intrinsic importance, worth or value;
- positive regard and esteem for other people as equal human beings who share a common dignity and have exactly the same set of human rights and freedoms irrespective of their particular cultural affiliations, beliefs, opinions, lifestyles or practices;
- 3. positive regard and esteem for the beliefs, opinions, lifestyles and practices adopted by other people, as long as these do not undermine or violate the dignity, human rights or freedoms of others.

<sup>8.</sup> From a human rights perspective, another person's right to freedom of beliefs should always be respected, but respect cannot be accorded to the contents of beliefs that seek to undermine or violate the dignity, human rights and fundamental freedoms of others. In the case of beliefs where the content cannot be respected, restrictions are placed not on the right to hold the beliefs but on the freedom to manifest those beliefs if such restrictions are necessary for public safety, the protection of public order, or the protection of the rights and freedoms of others (see Article 9 of the European Convention of Human Rights: www.echr.coe.int/Documents/Convention\_ENG.pdf).

# **Civic-mindedness**

Civic-mindedness is an attitude towards a community or social group. The term "community" is used here to denote a social or cultural group that is larger than one's immediate circle of family and friends and to which one feels a sense of belonging. There are numerous types of group that might be relevant here, for example, the people who live within a particular geographical area (such as a neighbourhood, a town or city, a country, a group of countries such as Europe or Africa, or indeed the world in the case of the "global community"), a more geographically diffused group (such as an ethnic group, faith group, leisure group, sexual orientation group, etc.), or any other kind of social or cultural group to which an individual feels a sense of belonging. Every individual belongs to multiple groups, and an attitude of civic-mindedness may be held towards any number of these. Civic-mindedness involves:

- 1. a feeling of belonging to and identification with the community;
- 2. mindfulness of other people in the community, of the interconnectedness between those people, and of the effects of one's actions on those people;
- a sense of solidarity with other people in the community, including a willingness to co-operate and work with them, feelings of concern and care for their rights and welfare, and a willingness to defend those who might be disempowered and disadvantaged within the community;
- 4. an interest in, and attentiveness towards, the affairs and concerns of the community;
- a sense of civic duty, a willingness to contribute actively to community life, a willingness to participate in decisions concerning the affairs, concerns and common good of the community, and a willingness to engage in dialogue with other members of the community regardless of their cultural affiliations;
- 6. a commitment to fulfil, to the best of one's abilities, the responsibilities, duties or obligations that are attached to the roles or positions which one occupies within the community;
- 7. a sense of accountability to other people within the community and accepting that one is answerable to others for one's decisions and actions.

# Responsibility

The term "responsibility" has many meanings. Two meanings that are especially pertinent to a culture of democracy are role responsibility and moral responsibility. The former is an aspect of civic-mindedness (see above, point 6); here we are concerned with the latter. Moral responsibility is an attitude towards one's own actions. It arises when a person has an obligation to act in a particular way and deserves praise or blame for either performing that act or failing to act in that way. Necessary conditions for individuals to be judged as being either praiseworthy or blameworthy are that they are able to reflect on their own actions, are able to form intentions about how they will act, and are able to execute their chosen actions (hence, when a lack of resources or structural conditions conspire to prevent a person from performing an action, it is inappropriate to ascribe either praise or blame to them). Responsibility can require courage insofar as taking a principled stance may entail acting on one's

own, taking action against the norms of a community, or challenging a collective decision that is judged to be wrong. Thus, there can sometimes be a tension between civic-mindedness (construed as solidarity with and loyalty towards other people) and moral responsibility. An attitude of responsibility for one's own actions therefore involves:

- 1. the adoption of a reflective and thoughtful approach towards one's actions and the possible consequences of those actions;
- 2. the identification of one's duties and obligations and how one ought to act in relation to a particular situation, based on a value or set of values;<sup>9</sup>
- 3. making decisions about the actions to take (which in some cases might entail not taking action), given the circumstances which apply;
- 4. the taking of action (or the avoidance of action) accordingly as an autonomous agent;
- 5. willingness to hold oneself accountable for the nature or consequences of one's decisions and actions;
- 6. willingness to appraise and judge the self;
- 7. willingness to act courageously when this is judged to be necessary.

# **Self-efficacy**

Self-efficacy is an attitude towards the self. It involves a positive belief in one's own ability to undertake the actions which are required to achieve particular goals. This belief commonly entails the further beliefs that one can understand what is required, can make appropriate judgments, can select appropriate methods for accomplishing tasks, can navigate obstacles successfully, can influence what happens, and can make a difference to the events that affect one's own and other people's lives. Thus, self-efficacy is associated with feelings of self-confidence in one's own abilities. Low self-efficacy can discourage democratic and intercultural behaviour even when there is a high level of ability, while unrealistically high self-efficacy can lead to frustration and disappointment. An optimal attitude is relatively high self-efficacy coupled to a realistically estimated high level of ability, which encourages individuals to tackle new challenges and enables them to take action on issues of concern. Thus, self-efficacy involves:

- 1. belief in one's ability to understand issues, to make judgments and to select appropriate methods for accomplishing tasks;
- 2. belief in one's ability to organise and execute the courses of action required to attain particular goals, and to navigate the obstacles that might arise;
- 3. a feeling of confidence about tackling new challenges;
- 9. Hence, the deployment of the attitude of responsibility in democratic and intercultural situations requires the simultaneous deployment of one or more of the three sets of values specified by the current model (i.e. valuing human dignity and human rights, valuing cultural diversity, or valuing democracy, justice, fairness, equality and the rule of law). In the absence of the simultaneous deployment of one or more of these sets of values, responsibility would not be a democratic competence but a more general political competence instead (cf. the introductory text on values, Chapter 6.1).

- 4. a feeling of confidence about democratic engagement and undertaking the actions judged to be necessary to achieve democratic goals (including challenging and holding to account those in positions of power and authority when their decisions or actions are judged to be unfair or unjust);
- 5. a feeling of confidence about engaging in intercultural dialogue with those who are perceived to have cultural affiliations that differ from one's own.

# **Tolerance of ambiguity**

Tolerance of ambiguity is an attitude towards objects, events and situations which are perceived to be uncertain and subject to multiple conflicting or incompatible interpretations. People who have high tolerance of ambiguity evaluate these kinds of objects, events and situations in a positive manner, willingly accept their inherent lack of clarity, are willing to admit that other people's perspectives may be just as adequate as their own perspectives, and deal with the ambiguity constructively. Hence, the term "tolerance" should be understood here in its positive sense of accepting and embracing ambiguity (rather than in its negative sense of enduring or putting up with ambiguity). People who have low tolerance of ambiguity instead adopt a single perspective on unclear situations and issues, hold a closed attitude towards unfamiliar situations and issues, and use fixed and inflexible categories for thinking about the world. Thus, in the present context, tolerance of ambiguity involves:

- 1. recognition and acknowledgement that there can be multiple perspectives on and interpretations of any given situation or issue;
- 2. recognition and acknowledgement that one's own perspective on a situation may be no better than other people's perspectives;
- 3. acceptance of complexity, contradictions and lack of clarity;
- 4. willingness to undertake tasks when only incomplete or partial information is available;
- 5. willingness to tolerate uncertainty and to deal with it constructively.

# 6.3. Skills

A skill is the capacity for carrying out complex, well-organised patterns of either thinking or behaviour in an adaptive manner in order to achieve a particular end or goal.

There are eight sets of skills that are important for a culture of democracy, as follows.

# **Autonomous learning skills**

Autonomous learning skills are those skills that individuals require to pursue, organise and evaluate their own learning, in accordance with their own needs, in a self-directed and self-regulated manner, without being prompted by others. Autonomous learning skills are important for a culture of democracy because they enable individuals to learn for themselves about, and how to deal with, political, civic and cultural issues using multiple and diverse sources both far and near, rather than relying on agents in their immediate environment for the provision of information about these issues. Autonomous learning skills include abilities or skills in:

- 1. identifying one's own learning needs these needs may stem from gaps in knowledge or understanding, from lack or poor mastery of skills, or from difficulties that have arisen as a consequence of current attitudes or values;
- identifying, locating and accessing possible sources of the information, advice or guidance which is required to address these needs – these sources could include personal experiences, interactions and discussions with others, encounters with people who are perceived to have different cultural affiliations from one's own or who hold different beliefs, opinions or world views from one's own, and visual, print, broadcast and digital media sources;
- 3. judging the reliability of the various sources of information, advice or guidance, assessing them for possible bias or distortion, and selecting the most suitable sources from the range available;
- 4. processing and learning the information, using the most appropriate learning strategies and techniques, or adopting and following the advice or guidance, from the most reliable sources, making adjustments to one's existing repertoire of knowledge, understanding, skills, attitudes or values accordingly;
- thinking about what has been learned, the progress that has been made, evaluating the learning strategies that have been used, and drawing conclusions about further learning that may still need to be undertaken and new learning strategies that may need to be acquired.

# **Analytical and critical thinking skills**

Analytical and critical thinking skills consist of a large and complex cluster of interrelated skills. Analytical thinking skills are those skills that are required to analyse materials of any kind (e.g. texts, arguments, interpretations, issues, events, experiences) in a systematic and logical manner. They include abilities or skills in:

1. systematically breaking down the materials that are under analysis into constituent elements, and organising those elements in a logical manner;

- 2. identifying and interpreting the meaning(s) of each element, possibly by comparing and relating those elements to what is already known and identifying similarities and differences;
- 3. examining the elements in relationship to each other and identifying the connections that exist between them (logical, causal, temporal, etc.);
- 4. identifying any discrepancies, inconsistencies or divergences between elements;
- 5. identifying alternative possible meanings and relationships for individual elements, generating new elements that may be missing from the whole, systematically changing elements to determine effects on the whole, and generating new syntheses of the elements that have been examined in other words, imagining and exploring novel possibilities and alternatives;
- 6. drawing the results of the analysis together in an organised and coherent manner to construct logical and defensible conclusions about the whole.

Critical thinking skills consist of those skills that are required to evaluate and make judgments about materials of any kind. They therefore include abilities or skills in:

- 1. making evaluations on the basis of internal consistency, and on the basis of consistency with available evidence and experience;
- 2. making judgments about whether or not materials under analysis are valid, accurate, acceptable, reliable, appropriate, useful and/or persuasive;
- 3. understanding and evaluating the preconceptions, assumptions and textual or communicative conventions upon which materials are based;
- 4. engaging not only with the literal meaning of materials, but also with their broader rhetorical purpose including the underlying motives, intentions and agendas of those who produced or created them (in the case of political communications, this includes the ability to identify propaganda and the ability to deconstruct the underlying motives, intentions and purposes of those who have produced the propaganda);
- situating the materials within the historical context in which they have been produced in order to assist in making evaluative judgments about the materials;
- 6. generating and elaborating different alternative options, possibilities and solutions to those that are present within the materials under consideration;
- 7. weighing up the pros and cons of the available options this can include cost-benefit analysis (incorporating both short-term and long-term perspectives), resource analysis (assessing whether the resources required for each option are available in practice) and risk analysis (understanding and assessing the risks associated with each option and how they might be managed);
- drawing the results of the evaluative process together in an organised and coherent manner to construct a logical and defensible argument for or against a particular interpretation, conclusion or course of action, based on explicit and specifiable criteria, principles or values and/or compelling evidence;

9. recognising one's own assumptions and preconceptions that might have biased the evaluative process, and acknowledging that one's beliefs and judgments are always contingent and dependent upon one's own cultural affiliations and perspective.

Effective analytical thinking incorporates critical thinking (i.e. the evaluation of the materials under analysis), while effective critical thinking incorporates analytical thinking (i.e. drawing distinctions and making connections). For this reason, analytical and critical thinking skills are inherently linked together.

# Skills of listening and observing

Skills of listening and observing are the skills that are required to understand what other people are saying and to learn from other people's behaviour. Understanding what other people are saying requires active listening – paying close attention not only to what is being said but also to how it is being said through the use of tone, pitch, loudness, rate and fluency of voice, and paying close attention to the person's accompanying body language, especially their eye movements, facial expressions and gestures. Close observational scrutiny of other people's behaviour can also be an important source of information about the behaviours that are most appropriate and effective in different social settings and cultural contexts, and can assist a learner in mastering those behaviours through the retention of that information and replicating the other person's behaviour in later similar situations. Thus, skills of listening and observing include abilities or skills in:

- 1. attending not only to what is being said but also to how it is being said and to the body language of the speaker;
- 2. attending to possible inconsistencies between verbal and non-verbal messages;
- 3. attending to subtleties of meaning and to what might be only partially said or indeed left unsaid:
- 4. attending to the relationship between what is being said and the social context in which it is said;
- 5. paying close attention to the behaviour of other people and retaining information about that behaviour, particularly the behaviour of others who are perceived to have different cultural affiliations from one's own;
- 6. paying close attention to the similarities and the differences in how people react to the same situation, particularly people who are perceived to have different cultural affiliations from one another.

# **Empathy**

Empathy is the set of skills required to understand and relate to other people's thoughts, beliefs and feelings, and to see the world from other people's perspectives. Empathy involves the ability to step outside one's own psychological frame of reference (i.e. to decentre from one's own perspective) and the ability to imaginatively apprehend and understand the psychological frame of reference and perspective

of another person. This skill is fundamental to imagining the cultural affiliations, world views, beliefs, interests, emotions, wishes and needs of other people. There are several different forms of empathy that can be distinguished, including:

- 1. cognitive perspective-taking the ability to apprehend and understand the perceptions, thoughts and beliefs of other people;
- 2. affective perspective-taking the ability to apprehend and understand the emotions, feelings and needs of other people;
- sympathy, sometimes called "compassionate empathy" or "empathic concern"

   the ability to experience feelings of compassion and concern for other people based on the apprehension of their cognitive or affective state or condition, or their material situation or circumstances.

# Flexibility and adaptability

Flexibility and adaptability are the skills that are required to adjust one's thoughts, feelings or behaviours in a principled manner to new contexts and situations so that one can respond effectively and appropriately to their challenges, demands and opportunities. Flexibility and adaptability enable individuals to adjust positively to novelty and change and to other people's social or cultural expectations, communication styles and behaviours. They also enable individuals to adjust their patterns of thinking, feeling or behaviour in response to new situational contingencies, experiences, encounters and information. Flexibility and adaptability, defined in this way, need to be distinguished from the unprincipled or opportunistic adjustment of behaviour for personal benefit or gain. They also need to be distinguished from externally coerced adaptation. Thus, flexibility and adaptability include abilities or skills in:

- 1. adjusting one's habitual way of thinking due to changing circumstances, or temporarily shifting into a different cognitive perspective in response to cultural cues;
- 2. reconsidering one's own opinions in the light of new evidence and/or rational argument;
- 10. Note the positioning of empathy as a skill in the current model. The term "empathy" is of course also used in many other ways in everyday discourse. For example, it is sometimes used when a person experiences the same emotion that another person is feeling (the phenomenon of "emotional contagion", where a person "catches" and shares another person's joy, panic, fear, etc.), sometimes to refer to a sense of emotional connectedness or identification with another person (e.g. "I had a lot of empathy for the leading character in the book") and sometimes to refer to the compassion or concern for another person that results from sympathy (e.g. "I feel empathy for you in your current predicament"). The term "empathy" is also sometimes used to refer to a much larger cluster of responses that one may have to another person in which openness to the other, respect for the other, cognitive and emotional engagement with the other, and feelings of emotional connection to the other are co-mingled. The present model instead uses the term "empathy" in a more specific and focused manner to denote the set of skills that are required to understand and relate to other people's thoughts, beliefs and feelings, this being a crucial set of skills for participating in a culture of democracy. This definition is not intended to preclude the possible simultaneous mobilisation and deployment of empathy, openness, respect, etc., as an entire cluster of competences or capacities in some situations.
- 11. For example, the enforced assimilation of cultural minorities to a majority culture should never be condoned. All individuals have a fundamental right to choose their own cultural affiliations, beliefs and lifestyle (see footnote 8).

- 3. controlling and regulating one's own emotions and feelings in order to facilitate more effective and appropriate communication and co-operation with others;
- 4. overcoming anxieties, worries and insecurities about meeting and interacting with other people who are perceived to have different cultural affiliations from one's own:
- 5. regulating and reducing negative feelings towards members of another group with which one's own group has historically been in conflict;
- 6. adjusting one's behaviour in a socially appropriate way according to the prevailing cultural environment;
- 7. adapting to different communication styles and behaviours, and switching to appropriate communication styles and behaviours to avoid violating the cultural norms of others and to communicate with them through means which they are able to understand.

# Linguistic, communicative and plurilingual skills<sup>12</sup>

Linguistic, communicative and plurilingual skills are those skills that are required to communicate effectively and appropriately with other people. They include the following abilities and skills, among others:<sup>13</sup>

- the ability to communicate clearly in a range of situations this includes expressing one's beliefs, opinions, interests and needs, explaining and clarifying ideas, advocating, promoting, arguing, reasoning, discussing, debating, persuading and negotiating;
- 2. the ability to meet the communicative demands of intercultural situations by using more than one language or language variety or by using a shared language or lingua franca to understand another language;
- 3. the ability to express oneself confidently and without aggression, even in situations where one is disadvantaged through a disparity of power, and to express a fundamental disagreement with another person in a manner that is nevertheless respectful of that person's dignity and rights;
- 4. the ability to recognise the different forms of expression and the different communicative conventions (both verbal and non-verbal) in the communications employed by other social groups and their cultures;
- 12. The term "language" is used in this document to denote all linguistic systems, whether recognised as languages or considered to be varieties of recognised languages, irrespective of modality. It includes spoken and signed language and all other forms of non-spoken language. The terms "verbal" and "non-verbal" communication in this context therefore mean, respectively, "communication effected by means of language" and "communication effected by means other than language".
- 13. First and foremost, of course, effective and appropriate communication requires linguistic skills (to produce and comprehend spoken and written sentences and utterances), sociolinguistic skills (to process accent, dialect, register and the linguistic markers of social relations between speakers) and discourse skills (to construct longer coherent stretches of language through the use of appropriate communicative conventions, and to deploy spoken discourse and written texts for particular communicative purposes). However, these are considered to be generic skills in the present context and are therefore omitted from the model. Readers who are interested in a detailed account of these particular skills should instead consult the Common European framework of reference for languages (see footnote 5), where they are described at length.

- the ability to adjust and modify one's communicative behaviour so that one
  uses the communicative conventions (both verbal and non-verbal) that are
  appropriate to one's interlocutor(s) and to the prevailing cultural setting;
- 6. the ability to ask questions of clarification in an appropriate and sensitive manner in cases where the meanings being expressed by another person are unclear or where inconsistencies between the verbal and non-verbal messages produced by another person are detected;
- 7. the ability to manage breakdowns in communication, for example by requesting repetitions or reformulations from others, or providing restatements, revisions or simplifications of one's own misunderstood communications;
- 8. the ability to act as a linguistic mediator in intercultural exchanges, including skills in translating, interpreting and explaining, and to act as an intercultural mediator by assisting others to understand and appreciate the characteristics of someone or something that is perceived to have a different cultural affiliation from their own.

# **Co-operation skills**

Co-operation skills are those skills that are required to participate successfully with others on shared activities, tasks and ventures. They include abilities or skills in:

- 1. expressing views and opinions in group settings, and encouraging other group members to express their views and opinions in such settings;
- 2. building consensus and compromise within a group;
- 3. taking action together with others in a reciprocal and co-ordinated manner;
- 4. identifying and setting group goals;
- 5. pursuing the goals of a group and adapting one's own behaviour for the purpose of achieving these goals;
- 6. appreciating all group members' talents and strengths, and helping others to develop in areas where they need to and want to improve;
- 7. encouraging and motivating other group members to co-operate and help each other in order to achieve group goals;
- 8. helping others with their work where appropriate;
- 9. sharing relevant and useful knowledge, experience or expertise with the group and persuading other group members to do so;
- 10. recognising conflict in group settings, including identifying emotional signs of conflict in the self and in others, and responding appropriately using peaceful means and dialogue.

# **Conflict-resolution skills**

Conflict-resolution skills are those skills that are required to address, manage and resolve conflicts in a peaceful way. They include abilities or skills in:

- 1. reducing or preventing aggression and negativity, and creating a neutral environment in which people feel free to express their differing opinions and concerns without fear of reprisal;
- 2. encouraging and enhancing receptivity, mutual understanding and trust between conflicting parties;
- 3. recognising differences in the power and/or status of the conflicting parties, and taking steps to reduce the possible impact of such differentials on communications between them;
- 4. effectively managing and regulating emotions the ability to interpret one's own underlying emotional and motivational states as well as those of others, and to deal with emotional stress, anxiety and insecurity both in oneself and in others;
- 5. listening to and understanding the different perspectives of the parties involved in conflicts;
- 6. expressing and summarising the different points of view held by conflicting parties;
- 7. countering or reducing misperceptions held by conflicting parties;
- 8. recognising that sometimes there may be a need for a period of silence, a truce or a period of inaction, to allow the conflicting parties to reflect on the perspectives that are held by others;
- 9. identifying, analysing, relating and contextualising the causes and other aspects of conflicts;
- 10. identifying common ground on which agreement between conflicting parties can be built, identifying options for resolving conflicts, and refining possible compromises or solutions;
- 11. assisting others to resolve conflicts by enhancing their understanding of the available options;
- 12. assisting and guiding the parties involved to agree on an optimal and acceptable solution to the conflict.

# 6.4. Knowledge and critical understanding

Knowledge is the body of information that is possessed by a person, while understanding is the comprehension and appreciation of meanings. The term "critical understanding" is used in the present context to emphasise the need for the comprehension and appreciation of meanings in the context of democratic processes and intercultural dialogue to involve active reflection on and critical evaluation of that which is being understood and interpreted (as opposed to automatic, habitual and unreflective interpretation).

The various forms of knowledge and critical understanding that are required for a culture of democracy fall into three main sets, as follows.

# Knowledge and critical understanding of the self

Self-awareness and self-understanding are vital for participating effectively and appropriately in a culture of democracy. Knowledge and critical understanding of the self has many different aspects, including:

- 1. knowledge and understanding of one's own cultural affiliations;
- 2. knowledge and understanding of one's perspective on the world and of its cognitive, emotional and motivational aspects and biases;
- 3. knowledge and understanding of the assumptions and preconceptions which underlie one's perspective on the world;
- 4. understanding how one's perspective on the world, and one's assumptions and preconceptions, are contingent and dependent upon one's cultural affiliations and experiences, and in turn affect one's perceptions, judgments and reactions to other people;
- 5. awareness of one's own emotions, feelings and motivations, especially in contexts involving communication and co-operation with other people;
- 6. knowledge and understanding of the limits of one's own competence and expertise.

# Knowledge and critical understanding of language and communication

Knowledge and critical understanding of language and communication have many different aspects, and include:

- 1. knowledge of the socially appropriate verbal and non-verbal communicative conventions which operate in the language(s) which one uses;
- 2. understanding that people of other cultural affiliations may follow different verbal and non-verbal communicative conventions from oneself, which are meaningful from their perspective, even when they are using the same language as oneself;
- 3. understanding that people who have different cultural affiliations can perceive the meanings of communications in different ways;

- 4. understanding that there are multiple ways of speaking in any given language and a variety of ways of using the same language;
- 5. understanding how the use of language is a cultural practice which operates as a carrier of information, meanings and identities which circulate in the culture in which that language is embedded;
- 6. understanding of the fact that languages may express culturally shared ideas in a unique way or express unique ideas which may be difficult to access through another language;
- 7. understanding the social impact and effects on others of different communication styles, including understanding how different communication styles may clash or result in a breakdown of communication;
- 8. understanding how one's own assumptions, preconceptions, perceptions, beliefs and judgments are related to the specific language(s) which one speaks.

# Knowledge and critical understanding of the world (including politics, law, human rights, culture, cultures, religions, history, media, economies, the environment and sustainability)

Knowledge and critical understanding of the world subsumes a large and complex range of knowledge and understanding in a variety of domains, including all of the following.

# (a) Knowledge and critical understanding of politics and law, which includes:

- knowledge and understanding of political and legal concepts, including democracy, freedom, justice, equality, citizenship, rights and responsibilities, the necessity of laws and regulations, and the rule of law;
- 2. knowledge and understanding of democratic processes, of how democratic institutions work, including the roles of political parties, election processes and voting;
- knowledge and understanding of the diverse ways in which citizens can
  participate in public deliberations and decision making and can influence
  policy and society, including understanding of the role that civil society
  and NGOs can play in this regard;
- 4. understanding power relations, political disagreement and conflict of opinion in democratic societies, and of how such disagreements and conflicts can be peacefully resolved;
- 5. knowledge and understanding of current affairs, contemporary social and political problems, and the political views of others;
- 6. knowledge and understanding of contemporary threats to democracy.

# (b) Knowledge and critical understanding of human rights, which includes:

1. knowledge and understanding that human rights are grounded in the dignity that is inherent in all human beings;

- knowledge and understanding that human rights are universal, inalienable and indivisible, and that everyone does not only have human rights but also has a responsibility to respect the rights of others, irrespective of their national origins, ethnicity, race, religion, language, age, sex, gender, political opinion, birth, social origin, property, disability, sexual orientation or other status;
- 3. knowledge and understanding of the obligations of states and governments in relation to human rights;
- 4. knowledge and understanding of the history of human rights, including the Universal Declaration of Human Rights, the European Convention on Human Rights, and the United Nations Convention on the Rights of the Child;
- 5. knowledge and understanding of the relationship between human rights, democracy, freedom, justice, peace and security;
- 6. knowledge and understanding that there may be different ways of interpreting and experiencing human rights in different societies and cultures but that the possible variations are framed by internationally agreed legal instruments which set out minimum standards for human rights irrespective of cultural context;
- 7. knowledge and understanding of how human rights principles are applied in practice to specific situations, how violations of human rights can arise, how violations of human rights can be addressed, and how possible conflicts between human rights can be resolved;
- 8. knowledge and understanding of critical human rights challenges in the world today.

# (c) Knowledge and critical understanding of culture and cultures, which includes:

- 1. knowledge and understanding of how people's cultural affiliations shape their world views, preconceptions, perceptions, beliefs, values, behaviours and interactions with others;
- knowledge and understanding that all cultural groups are internally variable
  and heterogeneous, do not have fixed inherent characteristics, contain
  individuals who contest and challenge traditional cultural meanings, and
  are constantly evolving and changing;
- knowledge and understanding of how power structures, discriminatory
  practices and institutional barriers within and between cultural groups
  operate to restrict opportunities for disempowered individuals;
- 4. knowledge and understanding of the specific beliefs, values, norms, practices, discourses and products that may be used by people who have particular cultural affiliations, especially those used by people with whom one interacts and communicates and who are perceived to have different cultural affiliations from oneself.

# (d) Knowledge and critical understanding of religions, which includes:

- knowledge and understanding of the key aspects of the history of particular religious traditions, of the key texts and key doctrines of particular religious traditions, and of the commonalities and differences which exist between different religious traditions;
- 2. knowledge and understanding of religious symbols, religious rituals and the religious uses of language;
- 3. knowledge and understanding of the key features of the beliefs, values, practices and experiences of individuals who practise particular religions;
- 4. understanding of the fact that the subjective experience and personal expressions of religions are likely to differ in various ways from the standard textbook representations of those religions;
- 5. knowledge and understanding of the internal diversity of beliefs and practices which exists within individual religions;
- knowledge and understanding of the fact that all religious groups contain individuals who contest and challenge traditional religious meanings, and that religious groups do not have fixed inherent characteristics, but are constantly evolving and changing.

# (e) Knowledge and critical understanding of history, which includes:

- 1. knowledge and understanding of the fluid nature of history and of how interpretations of the past vary over time and across cultures;
- knowledge and understanding of particular narratives from different perspectives about the historical forces and factors that have shaped the contemporary world;
- understanding of the processes of historical investigation, in particular of how facts are selected and constructed, and how they become evidence in the production of historical narratives, explanations and arguments;
- 4. understanding of the need to access alternative sources of information about history because the contributions of marginalised groups (e.g. cultural minorities and women) are often excluded from standard historical narratives;
- 5. knowledge and understanding of how histories are often presented and taught from an ethnocentric point of view;
- 6. knowledge and understanding of how the concepts of democracy and citizenship have evolved in different ways in different cultures over time;
- knowledge and understanding of how stereotyping is a form of discrimination that has been used to deny individuality and diversity to human beings and to undermine human rights, and in some cases has led to crimes against humanity;
- 8. understanding and interpreting the past in the light of the present with a view to the future, and understanding the relevance of the past to concerns and issues in the contemporary world.

# (f) Knowledge and critical understanding of the media, which includes:

- knowledge and understanding of the processes through which the mass media select, interpret and edit information before transmitting it for public consumption;
- knowledge and understanding of the mass media as commodities that involve producers and consumers, and of the possible motives, intentions and purposes that the producers of content, images, messages and advertisements for the mass media may have;
- knowledge and understanding of digital media, of how digital media content, images, messages and advertisements are produced, and of the various possible motives, intentions and purposes of those who create or reproduce them;
- 4. knowledge and understanding of the effects that mass media and digital media content can have on individuals' judgments and behaviours;
- 5. knowledge and understanding of how political messages, propaganda and hate speech in the mass media and digital media are produced, how these forms of communication can be identified, and how individuals can guard and protect themselves against the effects of these communications.

# (g) Knowledge and critical understanding of economies, the environment and sustainability, which includes:

- knowledge and understanding of economies and of the economic and financial processes that affect the functioning of society, including the relationship between employment, earnings, profit, taxation and government expenditure;
- 2. knowledge and understanding of the relationship between income and expenditure, the nature and consequences of debt, the real cost of loans, and the risk of loans beyond repayment capacity;
- 3. knowledge and understanding of the economic interdependence of the global community and of the impact that personal choices and patterns of consumption may have in other parts of the world;
- 4. knowledge and understanding of the natural environment, the factors that can impact on it, the risks associated with environmental damage, current environmental challenges, and the need for responsible consumption and environmental protection and sustainability;
- 5. knowledge and understanding of the connections between economic, social, political and environmental processes, especially when viewed from a global perspective;
- 6. knowledge and understanding of the ethical issues associated with globalisation.

# 7. Conclusion

s stated at the outset of this document, the model that is presented here is an attempt to provide a description of the competences that need to be acquired by learners if they are to become effective engaged citizens and live peacefully together with others as equals in culturally diverse democratic societies. It is hoped that the current model will prove useful for educational decision making and planning and will assist in the harnessing of educational systems for the purpose of preparing learners for life as democratically and interculturally competent citizens.

It is also hoped that this model will enable educational systems to empower learners as autonomous social agents who are capable of choosing and pursuing their own goals in life within the framework that is provided by democratic institutions and respect for human rights. Several competences in the model are especially pertinent to this goal. For example, if learners develop an attitude of openness towards other cultures, beliefs, world views and practices, they will be willing to explore and investigate other perspectives and modes of life that lie beyond the traditional ones with which they have grown up, expanding the range of their experiences and their horizons; if they acquire autonomous learning skills, they will be able to learn independently about these new perspectives and modes of life and not be dependent solely upon information provided by others in their immediate environment; and if they acquire analytical and critical thinking skills, they will be able to subject to detailed scrutiny alternative perspectives and modes of life, and new information and ideas, and will be able to make their own evaluative judgments about whether or not they are acceptable or desirable. In addition, if young people learn to value human dignity and human rights, cultural diversity and democracy, then these values will be used as the foundation for all of their choices and actions, and they will willingly pursue their lives in a manner that respects the dignity and human rights of other people and the principles of democracy.

In short, equipping learners with the competences specified by the current model is an essential step which needs to be taken to empower them to choose and pursue their own goals within the framework that is provided by respect for human rights and democratic processes. Equipping them with these competences through the educational system, alongside taking action to tackle structural disadvantages and inequalities, is crucial to ensure the future health of our culturally diverse democratic societies and the empowerment and flourishing of all young people who live within them.

# Appendix A

# The sources of the competence schemes that were audited by the project

his appendix contains the sources of all the competence schemes that were audited by the project. Some of the sources listed below contain more than a single competence scheme, while others duplicate competence schemes contained in other sources. For this reason, the number of documents in this list (102) differs from the total number of schemes that were audited (101). Unless otherwise stated, all websites were accessed on 17 February 2016.

- Abbe A., Gulick L. M. V. and Herman J. L. (2007), Cross-cultural competence in army leaders: a conceptual and empirical foundation. Army Research Institute for the Behavioral and Social Sciences, Arlington, VA, www.deomi.org/ CulturalReadiness/documents/CCCompetenceArmy.pdf.
- 2. AFS Intercultural Programs (2013), AFS educational goals, www.afs.org/afs-and-intercultural-learning/afs-educational-goals/.
- 3. Audigier F. (2000), *Basic concepts and core competencies for education for democratic citizenship*, Council of Europe, Strasbourg, www.ibe.unesco.org/fileadmin/user\_upload/Curriculum/SEEPDFs/audigier.pdf.
- 4. Barrett M. (2013), "Intercultural competence: a distinctive hallmark of interculturalism?", in Barrett M. (ed.), *Interculturalism and multiculturalism: similarities and differences* (pp. 147-68), Council of Europe Publishing, Strasbourg.
- Barrett M. (2015), "An integrative model of political and civic participation: linking the macro, social and psychological levels of explanation", in Barrett M. and Zani B. (eds), *Political and civic engagement: multidisciplinary perspectives* (pp. 162-87), Routledge, London.
- 6. Barrett M. and Brunton-Smith I. (2014), "Political and civic engagement and participation: towards an integrative perspective", *Journal of Civil Society* 10(1), 5-28.
- 7. Barrett M., Byram M., Lázár I., Mompoint-Gaillard P. and Philippou S. (2014), Developing intercultural competence through education, Council of Europe Publishing, Strasbourg, www.coe.int/t/dg4/education/pestalozzi/Source/ Documentation/Pestalozzi3.pdf.

- 8. Beacco J.-C., Byram M., Cavalli M., Coste D., Cuenat M. E., Goullier F. and Panthier J. (2010), *Guide for the development and implementation of curricula for plurilingual and intercultural education*, Council of Europe, Strasbourg, www.coe.int/t/dg4/linguistic/Source/Source2010\_ForumGeneva/GuideEPI2010\_EN.pdf.
- 9. Bertelsmann Stiftung and Fondazione Cariplo (2008), *Intercultural competence the key competence in the 21st century?*, Bertelsmann Foundation Publishers, Gütersloh.
- 10. Boix Mansilla V. and Jackson A. (2011), *Educating for global competence:* preparing our youth to engage the world, Asia Society, New York, http://asiasociety.org/files/book-globalcompetence.pdf.
- Bologna Working Group on Qualifications Frameworks (2005), A framework for qualifications of the European Higher Education Area, Ministry of Science, Technology and Innovation, Copenhagen, www.ond.vlaanderen.be/hogeronderwijs/bologna/documents/050218\_QF\_EHEA.pdf.
- 12. Brander P., Cardenas C, de Vicente Abad J., Gomes R. and Taylor M. (2004), Education pack "All different – all equal", Council of Europe, Strasbourg, http://eycb.coe.int/edupack/.
- 13. Brander P., De Witte L., Ghanea N., Gomes R., Keen E., Nikitina A. and Pinkeviciute J. (2012), *Compass: manual for human rights education with young people*, Council of Europe Publishing, Strasbourg, www.coe.int/t/dg4/eycb/Source/Compass\_2012\_FINAL.pdf.
- 14. Brett P., Mompoint-Gaillard P. and Salema M. H. (2009), How all teachers can support citizenship and human rights education: a framework for the development of competences, Council of Europe Publishing, Strasbourg, www.coe. int/t/dg4/education/edc/Source/Resources/Pack/How all Teachers EN.pdf.
- 15. Byram M. (1997), *Teaching and assessing intercultural communicative competence*, Multilingual Matters, Clevedon.
- 16. Byram M. (2008), From foreign language education to education for intercultural citizenship, Multilingual Matters, Clevedon.
- 17. Byram M., Barrett M., Ipgrave J., Jackson R. and Méndez García M. C. (2009a), *Autobiography of intercultural encounters: context, concepts and theories,* Council of Europe Publishing, Strasbourg, www.coe.int/t/dg4/autobiography/source/aie\_en/aie\_context\_concepts\_and\_theories\_en.pdf.
- 18. Byram M., Barrett M., Ipgrave J., Jackson R. and Méndez García M. C. (2009b), Autobiography of intercultural encounters: notes for facilitators, Council of Europe Publishing, Strasbourg, www.coe.int/t/dg4/autobiography/Source/ AIE\_en/AIE\_notes\_for\_facilitators\_en.pdf.
- 19. Byram M., Gribkova B. and Starkey H. (2002), *Developing the intercultural dimension in language teaching: a practical introduction for teachers*, Council of Europe Publishing, Strasbourg, www.coe.int/t/dg4/linguistic/source/guide\_dimintercult\_en.pdf.
- 20. Campaign for the Civic Mission of Schools (CCMS) (2011), *Guardian of democracy: The Civic Mission of Schools*, Campaign for the Civic Mission of

- Schools, Silver Spring, MD, http://civicmission.s3.amazonaws.com/118/f0/5/171/1/Guardian-of-Democracy-report.pdf.
- 21. Candelier M., Camilleri-Grima A., Castellotti V., de Pietro J.-F., Lőrincz I., Meißner F. J., Noguerol A. and Schröder-Sura A. (2012), FREPA a framework of reference for pluralistic approaches: competences and resources, European Centre for Modern Languages/Council of Europe, Strasbourg, http://carap.ecml.at/.
- 22. Catteeuw P. (2012), *A framework of reference for intercultural competence*, FARO, Brussels, www.faronet.be/e-documenten/a-framework-of-reference-for-intercultural-competence-a-21st-century-flemish-experiment.
- 23. Čehajić S., Brown R. and Castano E. (2008), "Forgive and forget? Antecedents and consequences of intergroup forgiveness in Bosnia Herzegovina", *Political Psychology* 29(3), 351-67.
- 24. Čehajić S., Effron D., Halperin E., Liberman V. and Ross L. (2011), "Affirmation, acknowledgment of in-group responsibility, group-based guilt, and support for reparative measures", *Journal of Personality and Social Psychology* 101, 256-70.
- 25. Cohen-Chen S., Halperin E., Crisp R. J. and Gross J. J. (2014), "Hope in the Middle East: malleability beliefs, hope, and the willingness to compromise for peace", *Social Psychological and Personality Science* 5 (1), 67-75.
- 26. Connerley M. L. and Pedersen P. B. (2005), *Leadership in a diverse and multi-cultural environment: developing awareness, knowledge, and skills*, Sage, Thousand Oaks, CA.
- Conseil Superieur des Programmes, Ministère de l'éducation nationale, France (2014), Projet d'enseignement moral et civique, http://cache.media. education.gouv.fr/file/Organismes/32/8/CSP-Projet\_EMC\_337328.pdf.
- 28. Council of Europe (1996), EDC consultative meeting, Doc. DECS/CIT (96)1.
- 29. Council of Europe (2009), *Policies and practices for teaching sociocultural diversity competences*, Council of Europe Publishing, Strasbourg, www. coe.int/t/dg4/education/diversity/Competences\_en.asp.
- 30. Crosbie V. (2014), "Capabilities for intercultural dialogue", *Language and Intercultural Communication* 14(1), 91-107.
- 31. Deardorff D. K. (2006), "The identification and assessment of intercultural competence as a student outcome of internationalization at institutions of higher education in the United States", *Journal of Studies in International Education* 10, 241-66.
- 32. Department for Education, England, UK (2013), *Citizenship programmes of study: Key Stages 3 and 4, National Curriculum in England*, www.gov. uk/government/uploads/system/uploads/attachment\_data/file/239060/SECONDARY\_national\_curriculum\_-\_Citizenship.pdf.
- 33. Departamento de Educación, Gobierno de Navarra [Government of Navarra] (2012), Competencia social y ciudadana y coeducación: orientaciones para las materias de primaria y secundaria, http://coeducando.educacion.navarra.es/

- wp-content/uploads/2010/11/competencia\_social\_y\_ciudadana\_y\_coeducacion.pdf.
- 34. Departamento de Educación, Universidades e Investigación, Gobierno Vasco [Basque Government] (n.d.), *Las competencias básicas en el Sistema Educativo de la C.A.P.V.*, www.hezkuntza.ejgv.euskadi.net/r43-573/es/contenidos/informacion/dig\_publicaciones\_innovacion/es\_curricul/adjuntos/14\_curriculum\_competencias\_300/300002c\_Pub\_BN\_Competencias\_Basicas\_c. pdf.
- 35. EC-CoE Youth Partnership (2014), *Guidelines for intercultural dialogue in non-formal learning/education activities*. European Commission and Council of Europe, http://pjp-eu.coe.int/documents/1017981/8492613/ICD2014. pdf/73add476-4e8e-4522-8fb5-67df5e4ccfc8.
- 36. ELICIT (2010), *Reference framework of competences for teaching European citizenship literacy*, www.elicitizen.eu/tools-for-education/reference-framework.
- 37. European Consortium for Accreditation (2005), *Dublin Descriptors*, http://ecahe.eu/w/index.php/Dublin\_Descriptors.
- 38. European Parliament (2006), Recommendation 2006/962/EC of the European Parliament and of the Council of 18 December 2006 on key competences for lifelong learning. Official Journal of the European Union, L 394/10, 30.12.2006, http://europa.eu/legislation\_summaries/education\_training\_youth/lifelong\_learning/c11090\_en.htm and http://eur-lex.europa.eu/legal-content/EN/ALL/;jsessionid=5CpqTzDLm1sDXXGD82RBdwXQKRv1fJLbn9y8ngmS8 GNmyNv9ynVL!-975582138?uri=CELEX:32006H0962.
- 39. Eurydice (2012), *Citizenship Education in Europe*, Education, Audiovisual and Culture Executive Agency, P9 Eurydice, Brussels.
- 40. Fantini A. E. and Tirmizi A. (2006), *Exploring and assessing intercultural competence*, World Learning Publications, Paper 1, http://digitalcollections.sit.edu/worldlearning\_publications/1.
- 41. Flowers N., Brederode-Santos M. E., Claeys J., Fazah R., Schneider A. and Szelényi Z. (2009), *Compasito: manual on human rights education for children* (2nd edition), Council of Europe Publishing, Strasbourg, www.eycb.coe.int/compasito/pdf/Compasito%20EN.pdf.
- 42. Frau-Meigs D. (2007), General Rapporteur's report on the workshop on "Media literacy and human rights: education for sustainable democratic societies", Graz, Austria, 5-7 December 2007.
- 43. Göksel A., Gomes R., Keen E. and Brander P. (2010), *Mosaic: the training kit for Euro-Mediterranean youth work*, Council of Europe Publishing, Strasbourg, http://youth-partnership-eu.coe.int/youth-partnership/publications/T-kits/11/6577\_T-Kit\_11\_A4\_assemble.pdf.
- 44. Gollob R., Krapf P. and Weidinger W. (eds) (2010), Educating for democracy: background materials on democratic citizenship and human rights education for teachers, Council of Europe Publishing, Strasbourg, www. coe.int/t/dg4/education/edc/Source/Resources/Teachers\_manuals/ Educating\_democracy\_EN.pdf.

- 45. Gonzalez R., Manzi J. and Noor M. (2011), "Intergroup forgiveness and reparation in Chile: the role of identity and intergroup emotions", in Tropp L. and Mallett R. (eds), *Moving beyond prejudice reduction: pathways to positive intergroup relations* (pp. 221-40), American Psychological Association, Washington, DC.
- 46. Government of Andorra (2015), "Annex I Perfil de l'alumne a l'acabament de la segona ensenyança de l'Escola Andorrana", in Decret d'ordenament de l'educació bàsica obligatòria del sistema educatiu andorrà, *Butlletí Oficial del Principat d'Andorra* 17, 4 de març del 2015, www.bopa.ad/bopa/027017/Documents/ga27017062.pdf.
- 47. Government of Hungary (2011), Decree on the issue, introduction and implementation of the National Core Curriculum. Magyar Közlöny.
- 48. Government of Spain (2013), *Boletín Oficial del Estado* 295, Martes 10 de diciembre de 2013, Sec. I. Pág. 97858-97921, www.boe.es/boe/dias/2013/12/10/pdfs/BOE-A-2013-12886.pdf.
- 49. Grindal K. (1997), *EDC*: basic concepts and core competences: the approach in Norway, Council of Europe, Strasbourg.
- 50. Halperin E., Crisp R., Husnu S., Trzesniewski K. H., Dweck C. S. and Gross J. J. (2012), "Promoting intergroup contact by changing beliefs: group malleability, intergroup anxiety and contact motivation", *Emotion* 12, 1192-95.
- 51. Halperin E., Porat R., Tamir M. and Gross J. J. (2013), "Can emotion regulation change political attitudes in intractable conflict? From the laboratory to the field", *Psychological Science* 24, 106-111.
- 52. Halperin E., Russell G. A., Trzesniewski H. K., Gross J. J. and Dweck S. C. (2011), "Promoting the peace process by changing beliefs about group malleability", *Science* 333, 1767-69.
- 53. Hewstone M., Cairns E., Voci A., Hamberger J. and Niens U. (2006), "Intergroup contact, forgiveness, and experience of 'The Troubles' in Northern Ireland", *Journal of Social Issues* 62(1), 99-120.
- 54. Himmelmann G. (2003), "Zukunft, Fachidentität und Standards der politischen Bildung", unpublished manuscript, TU Braunschweig, Institut für Sozialwissenschaften, Braunschweig [summarised in Byram, 2008].
- 55. Hoskins B. and Deakin Crick R. (2008), *Learning to learn and civic competences:* different currencies or two sides of the same coin? JRC Scientific and Technical Report EUR 23360 EN 2008, Office for Official Publications of the European Communities, Luxembourg.
- 56. Hoskins B., Villalba C. M. H. and Saisana M. (2012), The 2011 Civic Competence Composite Indicator (CCCI-2): measuring young people's civic competence across Europe based on the IEA International Citizenship and Civic Education Study, JRC Scientific and Technical Report EUR 25182 EN 2012, Office for Official Publications of the European Communities. Luxembourg.
- 57. Hoskins B., Villalba E., Van Nijlen D. and Barber C. (2008), Measuring civic competence in Europe: a composite indicator based on IEA Civic Education Study 1999 for 14 Years Old in School, JRC Scientific and Technical Report EUR

- 23210 EN 2008, Office for Official Publications of the European Communities, Luxembourg.
- 58. INCA (2004), INCA Framework (assessor version), https:/ec.europa.eu/migrant-integration/librarydoc/the-inca-project-intercultural-competence-assessment.
- 59. Jackson R. (2014), Signposts: policy and practice for teaching about religions and non-religious world views in intercultural education, Council of Europe Publishing, Strasbourg.
- 60. Karwacka-Vögele K. (2012), "Towards indicators for intercultural understanding" in Huber J. (ed.), *Intercultural competence for all: preparation for living in a heterogeneous world* (pp. 51-60), Council of Europe Publishing, Strasbourg.
- 61. Krammer R., Kühberger C. and Windischbauer E. (2008), *Die durch politische Bildung zu erwerbenden Kompetenzen: Ein Kompetenz-Strukturmodell*, Bundesministerium für Unterricht, Kunst und Kultur, Vienna, www.phsalzburg.at/fileadmin/PH\_Dateien/Politische\_Bildung\_Windischbauer/KompetenzmodellKurzUndLangversion.pdf.
- 62. Lázár I. (ed.) (2012), *Recognising intercultural competence: what shows that I am interculturally competent?* Council of Europe Publishing, Strasbourg, www.coe.int/t/dg4/education/Pestalozzi/home/What/ICCTool\_en.asp.
- 63. Mata P. and Del Olmo M. (2009), "What do we mean by intercultural education? Results of an analysis on what INTER Network member think IE is and should be put in practice", in Gil Jaunena I. (ed.), *Diversity, inclusion and the values of democracy: building teachers' competences for intercultural education* (pp. 12-23), European Commission's Lifelong Learning Programme: INTER Network, 134367-LLP-1-2007-1-ES-COMENIUS-CNW.
- 64. Messner W. and Schäfer N. (2012), "Advancing intercultural competencies for global collaboration", in Bäumer U., Kreutter P. and Messner W. (eds), *Globalization of professional services* (pp. 189-201), Springer Verlag, Berlin.
- 65. Milot M. (2007), "The religious dimension in intercultural education", in Keast J. (ed.), *Religious diversity and intercultural education: a reference book for schools* (pp. 19-30), Council of Europe Publishing, Strasbourg.
- 66. Mompoint-Gaillard P. and Lázár I. (eds) (2015), TASKs for democracy: 60 activities to learn and assess transversal attitudes, skills and knowledge, Council of Europe Publishing, Strasbourg.
- 67. National Task Force on Civic Learning and Democratic Engagement (NTF) (2012), *A crucible moment: college learning and democracy's future*, Association of American Colleges and Universities, Washington, DC.
- 68. Neuner G. (2012), "The dimensions of intercultural education", in Huber J. (ed.), *Intercultural competence for all: preparation for living in a heterogeneous world* (pp. 11-49), Council of Europe Publishing, Strasbourg.
- 69. Noor M. and Cantacuzino M. (2013), *The forgiveness toolbox*, www.theforgivenesstoolbox.com.

- 70. Noor M., Brown R. J. and Prentice G. (2008), "Precursors and mediators of intergroup reconciliation in Northern Ireland: a new model", *British Journal of Social Psychology* 47(3), 481-95.
- 71. Noor M., Brown R., Gonzalez R., Manzi J. and Lewis C. A. (2008), "On positive psychological outcomes: what helps groups with a history of conflict to forgive and reconcile with each other?", *Personality and Social Psychology Bulletin* 34(6), 819-32.
- 72. North-South Centre of the Council of Europe (2010), *Global education guidelines: a handbook for educators to understand and implement global education*, North-South Centre of the Council of Europe, Lisbon, www.coe. int/t/dg4/nscentre/GE/GE/GE-Guidelines/GEguidelines-web.pdf.
- 73. Nussbaum M. (1997), *Cultivating humanity: a classical defense of reform in liberal education*, Harvard University Press, Cambridge, MA.
- 74. Nussbaum M. (2006), "Education and democratic citizenship: capabilities and quality education", *Journal of Human Development* 7(3), 385-95.
- 75. O'Regan J. P. and MacDonald M. N. (2007), Consultation report on the development of National Occupational Standards in intercultural working, CILT, The National Centre for Languages, London.
- 76. Organisation for Economic Co-operation and Development (OECD) (2005), *The definition and selection of key competences: executive summary.* Organisation for Economic Co-operation and Development/DeSeCo, Paris. www.oecd. org/pisa/35070367.pdf.
- 77. OSCE/ODIHR (2012), Guidelines on human rights education for secondary school systems, OSCE, Warsaw, www.osce.org/odihr/93969.
- 78. Oxfam (2006), Education for global citizenship: a guide for school, Oxfam, Oxford, www.oxfam.org.uk/~/media/Files/Education/Global%20Citizenship/education\_for\_global\_citizenship\_a\_guide\_for\_schools.ashx.
- 79. Partnership for 21st Century Skills (2009), *P21 framework definitions*, www. p21.org/storage/documents/P21\_Framework\_Definitions.pdf.
- 80. Portera A. (2014), "Intercultural competence in education, counselling and psychotherapy", *Intercultural Education* 25, 157-74.
- 81. Prats E., Tey A. and Martínez M. (2011), *Educar per una ciutadania activa a l'escola*, Aula de Ciutadania, Barcelona.
- 82. Ruffino R. (2012), "Intercultural education and pupil exchanges", in Huber J. (ed.), *Intercultural competence for all: preparation for living in a heterogeneous world* (pp. 63-94), Council of Europe Publishing, Strasbourg.
- 83. Rychen D. S. and Salganik L. H. (eds) (2003), *Key competences for a successful life and a well-functioning society*, Hogrefe and Huber, Göttingen.
- 84. Saguy T. and Halperin E. (2014), "Exposure to outgroup members criticizing their own group facilitates intergroup openness", *Personality and Social Psychology Bulletin* 40, 791-802.
- 85. Salto-Youth Cultural Diversity Resource Centre (2012), *Intercultural competence: research report*, www.salto-youth.net/rc/cultural-diversity/publications/interculturalcompetenceresearchreport/.

- 86. Schulz W., Ainley J., Fraillon J., Kerr D. and Losito B. (2010a), *Initial findings from the IEA International Civic and Citizenship Education Study*, International Association for the Evaluation of Educational Achievement, Amsterdam, www.iea.nl/fileadmin/user\_upload/Publications/Electronic\_versions/ICCS\_2009\_Initial\_Findings.pdf.
- 87. Schulz W., Ainley J., Fraillon J., Kerr D. and Losito B. (2010b), *ICCS 2009 international report: civic knowledge, attitudes and engagement among low-er-secondary school students in 38 countries*, International Association for the Evaluation of Educational Achievement, Amsterdam, www.iea.nl/fileadmin/user\_upload/Publications/Electronic\_versions/ICCS\_2009\_International\_Report.pdf.
- 88. Schulz W., Ainley J. and van de Gaer E. (2010), "Preparedness for active citizenship among lower secondary students in international comparison". Paper presented at the 4th IEA International Research Conference, Gothenburg, 1-3 July 2010.
- 89. Shnabel N., Halabi S. and Noor M. (2013), "Overcoming competitive victim-hood and facilitating forgiveness through re-categorization into a common victim and perpetrator identity", *Journal of Experimental Social Psychology* 49, 867-77.
- 90. Soland J., Hamilton L. S. and Stecher B. M. (2013), *Measuring 21st century competencies: guidance for educators*, Asia Society, New York, http://asiasociety.org/files/gcen-measuring21cskills.pdf.
- 91. Stradling R. (2001), *Teaching 20th century European history*, Council of Europe Publishing, Strasbourg, www.sissco.it/download/dossiers/estradling.pdf.
- 92. Sue D. W. (2001), "Multidimensional facets of cultural competence", *The Counseling Psychologist* 29(6), 790-821.
- 93. Tam T., Hewstone M., Cairns E., Tausch N., Maio G. and Kenworthy J. (2006), "The impact of intergroup emotions on forgiveness in Northern Ireland", *Group Processes and Intergroup Relations* 10(1), 119-35.
- 94. Tam T., Hewstone M., Kenworthy J. B., Cairns E., Marinetti C., Geddes L. and Parkinson B. (2008), "Postconflict reconciliation: intergroup forgiveness and implicit biases in Northern Ireland", *Journal of Social Issues* 64(2), 303-20.
- 95. Tuning Educational Structures in Europe (n.d.-a), *Approaches to teaching, learning and assessment in competences based degree programmes*, www. unideusto.org/tuningeu/images/stories/archivos/TLA%20PARA%20LA%20 PAGINA.pdf.
- 96. Tuning Educational Structures in Europe (n.d.-b), *Generic competences*, www. unideusto.org/tuningeu/competences/generic.html.
- 97. Tuning Educational Structures in Europe (n.d.-c), *Education specific competences*, www.unideusto.org/tuningeu/competences/specific/education.
- 98. Ulrich M. and Oser F. (1999), *Participatory experience and individual competence*, Council of Europe Publishing, Strasbourg.

- 99. UNESCO (2013), Intercultural competences: conceptual and operational framework, UNESCO, Paris, http://unesdoc.unesco.org/images/0021/002197/219768e.pdf.
- 100. UNESCO (2014), *Global citizenship education: preparing learners for the challenges of the twenty-first century*, UNESCO, Paris, http://unesdoc.unesco.org/images/0022/002277/227729E.pdf.
- 101. Veldhuis R. (1997), Education for democratic citizenship: dimensions of citizenship, core competencies, variables and international activities, Doc DECS/CIT (97)23, Council of Europe, Strasbourg.
- 102. Zukin C., Keeter S., Andolina M., Jenkins K. and Delli Carpini M. X. (2006), *A new engagement? Political participation, civic life, and the changing American citizen*, Oxford University Press, New York.

# Appendix B

# The 55 possible competences that were identified across the 101 competence schemes

Note that this list includes only competences in the 101 schemes which were judged to be relevant to a culture of democracy. It excludes other competences in the schemes which are relevant to other contexts instead (e.g. employability).

- 1. Valuing/respect for other human beings
- 2. Valuing/respect for cultural difference and diversity
- 3. Valuing/respect for human rights
- 4. Valuing/respect for democracy
- 5. Valuing/respect for justice, equity, fairness, equality and the rule of law
- 6. Valuing/respect for other cultures/cultural otherness, including attitudinal openness towards other cultures
- 7 Tolerance
- 8. Civic-mindedness
- 9. Responsibility
- 10. Self-efficacy and self-confidence
- 11. Autonomy and initiative
- 12. Tolerance of ambiguity
- 13. Capacity for co-operation and collaboration
- 14. Capacity to resolve conflict
- 15. Empathy and decentring
- 16. Multiperspectivity
- 17. Self-awareness and self-knowledge
- 18. Communicative awareness
- 19. Linguistic knowledge
- 20. Linguistic and communicative skills
- 21. Skills of listening and observing
- 22. Emotional awareness

- 23. Emotions and motivations to participate civically and politically
- 24. Stress-management, emotional regulation, self-regulation
- 25. Political and civic interest
- 26. Political and civic opinions and beliefs
- 27. Knowledge and understanding of politics and the law
- 28. Knowledge and understanding of financial matters and economies
- 29. Knowledge and understanding of human rights
- 30. Knowledge and understanding of specific cultures
- 31. Knowledge and understanding of culture in general
- 32. Knowledge and understanding of religions
- 33. Knowledge and understanding of the natural world and environmental sustainability
- 34. Knowledge and understanding of history
- 35. Knowledge and understanding of the present world
- 36. Media knowledge and skills (both traditional and new media)
- 37. Skills of knowledge discovery and of learning through interaction
- 38. Skills of interpreting and relating
- 39. Skills of general reasoning and analysis
- 40. The ability to differentiate between and assess long-term v. short-term benefits, advantages or goals
- 41. Critical thinking
- 42. Critical cultural awareness
- 43. Ethical/moral reasoning, judgment and integrity
- 44. Problem-solving
- 45. Decision making
- 46. Creativity
- 47. Flexibility and adaptability
- 48. Perseverance
- 49. Cognitive orientation or style
- 50. Attitudes to learning
- 51. Practical skills
- 52. Intercultural behaviour
- 53. Political participation
- 54. Civic participation
- 55. Specialised professional competences (e.g. of teachers, psychological counsellors)

# Appendix C

# Suggestions for further reading beyond the references listed in Appendix A<sup>14</sup>

- ► Ayton-Shenker D. (1995), *The challenge of human rights and cultural diversity*, United Nations Department of Public Information, DPI/1627/HR, New York.
- ► Bandura A. (1994), "Self-efficacy", in Ramachaudran V. S. (ed.), *Encyclopedia of Human Behaviour*, Vol. 4 (pp. 71-81), Academic Press, New York.
- ▶ Barrett M. (ed.) (2013), *Interculturalism and multiculturalism: similarities and differences*, Council of Europe Publishing, Strasbourg.
- Barrett M. and Zani B. (eds) (2015), Political and civic engagement: multidisciplinary perspectives, Routledge, London.
- ▶ Bertelsmann Foundation (2003), *Civic-mindedness: participation in modern society*, Bertelsmann Foundation Publishers, Gütersloh.
- ▶ Brady D., Verba S. and Schlozman K. L. (1995), "Beyond SES: a resource model of political participation", *American Political Science Review* 89, 271-94.
- ► Cantle T. (2012), *Interculturalism: the new era of cohesion and diversity*, Palgrave Macmillan, Basingstoke.
- Chiu C.-Y. and Hong Y.-Y. (2006), Social psychology of culture, Psychology Press, New York.
- ► Coste D. and Cavalli M. (2015), *Education, mobility, otherness: the mediation function of schools*, Language Policy Unit, Council of Europe, Strasbourg.
- ➤ Council of Europe (1950), European Convention on Human Rights, European Court of Human Rights, Council of Europe, Strasbourg, available at www.echr. coe.int/Documents/Convention\_ENG.pdf, accessed 17 February 2016.
- ➤ Council of Europe (2001), Common European framework of reference for languages: learning, teaching, assessment (CEFR), Cambridge University Press, Cambridge, available at www.coe.int/t/dg4/linguistic/source/framework\_en.pdf, accessed 17 February 2016.
- ➤ Council of Europe (2008), White Paper on intercultural dialogue: "Living together as equals in dignity", Committee of Ministers, Council of Europe Publishing, Strasbourg, available at www.coe.int/t/dg4/intercultural/source/white%20 paper\_final\_revised\_en.pdf, accessed 17 February 2016.

<sup>14.</sup> References to additional materials may be found in the annotated bibliography that has been produced by the current project (see www.coe.int/competences for further details).

- ➤ Council of Europe (2011), Living together: combining diversity and freedom in 21st century Europe. Report of the Group of Eminent Persons of the Council of Europe, Council of Europe Publishing, Strasbourg, available at www.coe.int/t/dg4/highereducation/2011/KYIV%20WEBSITE/Report%20on%20diversity.pdf, accessed 17 February 2016.
- Davis M. H. (1994), Empathy: a social psychological approach, Westview Press, Boulder. CO.
- ▶ Deardorff D. K. (ed.) (2009), *The SAGE handbook of intercultural competence*, Sage, Thousand Oaks, CA.
- ▶ Dillon R. S. (2014), "Respect", *Stanford Encyclopedia of Philosophy*, available at http://plato.stanford.edu/entries/respect, accessed 17 February 2016.
- ▶ Donnelly J. (2013), *Universal human rights in theory and practice* (3rd edition), Cornell University Press, Ithaca, NY.
- ▶ Eisenberg N. (2003), "Prosocial behavior, empathy, and sympathy", in Bornstein M. H., Davidson L., Keyes C. L. M. and Moore K. A. (eds) (2003), Well-being: positive development across the life course (pp. 253-65), Lawrence Erlbaum Associates Publishers, Mahwah, NJ.
- ▶ Eshleman A. (2014), "Moral responsibility", *Stanford Encyclopedia of Philosophy*, available at http://plato.stanford.edu/entries/moral-responsibility, accessed 17 February 2016.
- ► Feldmann E., Henschel T. R. and Ulrich S. (2000), *Tolerance: basis for democratic interaction*. Bertelsmann Foundation Publishers, Gütersloh.
- ► Furnham A. and Marks J. (2013), "Tolerance of ambiguity: a review of the recent literature", *Psychology* 4(9), 717-28.
- ► Hua C. W. and Wan K. E. (2011), *Civic mindedness: components, correlates and implications for the public service*, Civil Service College. Singapore.
- ► Klein M. (1995), "Responsibility", in Honderich T. (ed.), *The Oxford Companion to Philosophy* (pp. 771-2), Oxford University Press, Oxford.
- ► Knauth T. (2010), *Tolerance a key concept for dealing with cultural and religious diversity in education*, The European Wergeland Centre Statement Series, Oslo.
- ► Moore T. (2013), "Critical thinking: seven definitions in search of a concept", Studies in Higher Education 38(4), 506-22.
- ▶ Oakes P. J. Haslam S. A. and Turner J. C. (1994), *Stereotyping and social reality*, Blackwell, Oxford.
- Osler A. and Starkey H. (2005), Changing citizenship: democracy and inclusion in education, Open University Press, Maidenhead, Berkshire.
- ▶ Reber A. S. (1985), *The Penguin dictionary of psychology*, Penguin, Harmondsworth.
- Risager K. (2007), Language and culture pedagogy: from a national to a transnational paradigm, Multilingual Matters, Clevedon.
- Schwartz S. H. (1992), "Universals in the content and structure of values: theory and empirical tests in 20 countries", in Zanna M. (ed.), Advances in experimental social psychology (Vol. 25) (pp. 1-65), Academic Press, New York.

- ► Schwartz S. H. (2006), "Les valeurs de base de la personne: théorie, mesures et applications", *Revue française de sociologie* 42, 249-88.
- ► Sherrod L. R., Torney-Purta J. and Flanagan C. A. (eds) (2010), *Handbook of research on civic engagement in youth*, John Wiley & Sons, Hoboken, N&J.
- ► Stradling R. (2003), *Multiperspectivity in history teaching: a guide for teachers*, Council of Europe Publishing, Strasbourg.
- ▶ United Nations (2015), *Transforming our world: the 2030 agenda for sustainable development*, available at www.un.org/pga/wp-content/uploads/sites/3/2015/08/120815\_outcome-document-of-Summit-for-adoption-of-thepost-2015-development-agenda.pdf, accessed 17 February 2016.
- ▶ Verba S., Schlozman K. L. and Brady H. E. (1995), *Voice and equality: civic volunteerism in American politics*, Harvard University Press, Cambridge, MA.

# Appendix D

# **Authors and contributors**

# Main author:

► Martyn Barrett

## Contributions from:

Michael Byram, Luisa De Bivar Black, Jaroslav Faltyn, Claudia Lenz, Pascale Mompoint-Gaillard, Ketevan Natriashvili, Milica Popović, Câlin Rus, Salvador Sala, Hilligje van't Land, Natalia Voskresenskaya, Pavel Zgaga.

# **Council of Europe Education Department:**

► Sjur Bergan, Villano Qiriazi, Christopher Reynolds, Mireille Wendling.

# Acknowledgements:

The Council of Europe Education Department is very grateful to the following for reviewing and providing feedback on the draft model of Competences for Democratic Culture:

▶ Erik Amnå, Léonce Bekemans, Cesar Bîrzea, Lavinia Bracci, Fabiana Cardetti, Marisa Cavalli, Daniel Coste, Leah Davcheva, Darla Deardorff, Miguel Angel Garcia Lopez, Rolf Gollob, Richard Harris, Prue Holmes, Bryony Hoskins, Caroline Howarth, Ted Huddlestone, Andreas Körber, Ildikó Lázár, Rachel Lindner, Petr Macek, David Muller, Natia Natsvlishvili, Oana Nestian Sandu, Stavroula Philippou, Alison Phipps, Agostino Portera, Marzena Rafalska, Roberto Ruffino, Hugh Starkey, Olena Styslavska, Angela Tesileanu, Felisa Tibbitts, Judith Torney-Purta, Angelos Vallianatos, Manuela Wagner, Charlotte Wieslander, Robin Wilson, Ulrike Wolff-Jontofsohn, Fatmiroshe Xhemalaj.

The project to develop Competences for Democratic Culture has also benefited from the active and financial support of the ministries of education of Andorra, Georgia and Norway.

# Sales agents for publications of the Council of Europe Agents de vente des publications du Conseil de l'Europe

### BELGIUM/BELGIQUE

La Librairie Européenne -The European Bookshop Rue de l'Orme, 1 BE-1040 BRUXELLES Tel.: +32 (0)2 231 04 35 Fax: +32 (0)2 735 08 60 E-mail: info@libeurop.eu http://www.libeurop.be

Jean De Lannoy/DL Services Avenue du Roi 202 Koningslaan BE-1190 BRUXELLES Tel.: +32 (0)2 538 43 08 Fax: +32 (0)2 538 08 41 E-mail: jean.de.lannoy@dl-servi.com http://www.jean-de-lannoy.be

### BOSNIA AND HERZEGOVINA/ BOSNIE-HERZÉGOVINE

Robert's Plus d.o.o. Marka Maruliça 2/V BA-71000 SARAJEVO Tel.: + 387 33 640 818 Fax: + 387 35 640 818 E-mail: robertsplus@bih.net.ba

### CANADA

Renouf Publishing Co. Ltd.
22-1010 Polytek Street
CDN-OTTAWA, ONT K1J 9J1
Tel.: +1 613 745 7660
Toll-Free Tel.: (866) 767-6766
E-mail: order.dept@renoufbooks.com
http://www.renoufbooks.com

### CROATIA/CROATIE Robert's Plus d.o.o.

Marasoviçeva 67 HR-21000 SPLIT Tel.: + 385 21 315 800, 801, 802, 803

Fax: + 385 21 315 804 E-mail: robertsplus@robertsplus.hr

### CZECH REPUBLIC/ RÉPUBLIQUE TCHÈQUE

Suweco CZ, s.r.o. Klecakova 347 CZ-180 21 PRAHA 9 Tel.: +420 2 424 59 204 Fax: +420 2 848 21 646 E-mail: import@suweco.cz http://www.suweco.cz

# DENMARK/DANEMARK

OAD
Vimmelskaftet 32
DK-1161 KØBENHAVN K
Tel.: +45 77 66 60 00
Fax: +45 77 66 60 01
E-mail: reception@gad.dk
http://www.gad.dk

### FINLAND/FINLANDE

Akateeminen Kirjakauppa PO Box 128 Keskuskatu 1 FI-00100 HELSINKI Tel.: +358 (0)9 121 4430 Fax: +358 (0)9 121 4242 E-mail: akatilaus@akateeminen.com http://www.akateeminen.com

## FRANCE

Please contact directly / Merci de contacter directement Council of Europe Publishing Editions du Conseil de l'Europe FR-67075 STRASBOURG cedex Tel.: +33 (0)3 88 41 25 81 Fax: +33 (0)3 88 41 39 10 E-mail: publishing@coe.int http://book.coe.int

Librairie Kléber 1 rue des Francs-Bourgeois FR-67000 STRASBOURG Tel.: +33 (0)3 88 15 78 88 Fax: +33 (0)3 88 15 78 80 E-mail: librairie-kleber@coe.int http://www.librairie-kleber.com

### GREECE/GRÈCE

Librairie Kauffmann s.a. Stadiou 28 GR-105 64 ATHINAI Tel.: +30 210 32 55 321 Fax.: +30 210 32 30 320 E-mail: ord@otenet.gr http://www.kauffmann.gr

## HUNGARY/HONGRIE

Euro Info Service Pannónia u. 58. PF. 1039 HU-1136 BUDAPEST Tel.: +36 1 329 2170 Fax: +36 1 349 2053 E-mail: euroinfo@euroinfo.hu

### ITALY/ITALIE

Licosa SpA
Via Duca di Calabria, 1/1
IT-50125 FIRENZE
Tel.: +39 0556 483215
Fa-mail: licosa@licosa.com
http://www.licosa.com

# NORWAY/NORVÈGE

Akademika
Postboks 84 Blindern
NO-0314 OSLO
Tel.: 447 2 218 8100
Fax: +47 2 218 8103
E-mail: support@akademika.no
http://www.akademika.no

### POLAND/POLOGNE

Ars Polona JSC
25 Obroncow Street
PL-03-933 WARSZAWA
Tel.: +48 (0)22 509 86 00
Fax: +48 (0)22 509 86 10
E-mail: arspolona@arspolona.com.pl
http://www.arspolona.com.pl

### PORTUGAL

Marka Lda Rua dos Correeiros 61-3 PT-1100-162 LISBOA Tel: 351 21 3224040 Fax: 351 21 3224044 Web: www.marka.pt E mail: apoio.clientes@marka.pt

### RUSSIAN FEDERATION/ FÉDÉRATION DE RUSSIE

Ves Mir 17b, Butlerova ul. - Office 338 RU-117342 MOSCOW Tel.: +7 495 739 0971 Fax: +7 495 739 0971 E-mail: orders@vesmirbooks.ru http://www.vesmirbooks.ru

### SWITZERLAND/SUISSE

Planetis Sàrl 16 chemin des Pins CH-1273 ARZIER Tel.: +41 22 366 51 77 Fax: +41 22 366 51 78 E-mail: info@planetis.ch

### TAIWAN

Tycoon Information Inc.
5th Floor, No. 500, Chang-Chun Road
Taippi, Taiwan
Tel.: 886-2-8712 8886
Fax: 886-2-8712 4747, 8712 4777
E-mail: info@tycoon-info.com.tw
orders@tycoon-info.com.tw

## UNITED KINGDOM/ROYAUME-UNI

The Stationery Office Ltd PO Box 29 GB-NORWICH NR3 1GN Tel.: +44 (0)870 600 5522 Fax: +44 (0)870 600 5533 E-mail: book.enquiries@to.co.uk http://www.tsoshop.co.uk

# UNITED STATES and CANADA/ ÉTATS-UNIS et CANADA

Manhattan Publishing Co 670 White Plains Road USA-10583 SCARSDALE, NY Tel: + 1 914 472 4650 Fax: + 1 914 472 4316 E-mail: coe@manhattanpublishing.com http://www.manhattanpublishing.com

FR-67075 STRASBOURG Cedex

Contemporary societies within Europe face many challenges, including declining levels of voter turnout in elections, increased distrust of politicians, high levels of hate crime, intolerance and prejudice towards minority ethnic and religious groups, and increasing levels of support for violent extremism. These challenges threaten the legitimacy of democratic institutions and peaceful co-existence within Europe.

Formal education is a vital tool that can be used to tackle these challenges. Appropriate educational input and practices can boost democratic engagement, reduce intolerance and prejudice, and decrease support for violent extremism. However, to achieve these goals, educationists need a clear understanding of the democratic competences that should be targeted by the curriculum.

This book presents a new conceptual model of the competences which citizens require to participate in democratic culture and live peacefully together with others in culturally diverse societies. The model is the product of intensive work over a two-year period, and has been strongly endorsed in an international consultation with leading educational experts. The book describes the competence model in detail, together with the methods used to develop it. The model provides a robust conceptual foundation for the future development of curricula, pedagogies and assessments in democratic citizenship and human rights education. Its application will enable educational systems to be harnessed effectively for the preparation of students for life as engaged and tolerant democratic citizens. The book forms the first component of a new Council of Europe reference framework of competences for democratic culture. It is vital reading for all educational policy makers and practitioners who work in the fields of education for democratic citizenship, human rights education and intercultural education.

www.coe.int

The Council of Europe is the continent's leading human rights organisation. It comprises 47 member states, 28 of which are members of the European Union. All Council of Europe member states have signed up to the European Convention on Human Rights, a treaty designed to protect human rights, democracy and the rule of law. The European Court of Human Rights oversees the implementation of the Convention in the member states.



http://book.coe.int ISBN 978-92-871-8237-1 €13 / US\$26

